# Smart Agriculture: A Comprehensive Overview

Alakananda Mitra \* • Sukrutha L. T. Vangipuram • Anand K. Bapatla • Venkata K. V. V. Bathalapalli • Saraju P. Mohanty • Elias Kougianos • Chittaranjan Ray

the date of receipt and acceptance should be inserted later

Abstract The world population is anticipated to increase by 2 billion by 2050 causing a rapid escalation of food demand. A recent projection shows that the world is lagging behind in accomplishing the "Zero Hunger" goal, in spite of some advancements. Socio-economic and well-being fallout will affect food security. Vulnerable groups of people will suffer malnutrition. The agricultural industry must be upgraded, smartened, and automated to serve the growing population. Adopting existing technologies can make traditional agriculture efficient, sustainable, and eco-friendly. In this survey,

A. Mitra (Corresponding Author)
Nebraska Water Center at the Institute of Agriculture and Natural Resource, University of Nebraska-Lincoln, USA
E-mail: amitra6@unl.edu.
S. L. T. Vangipuram
Dept. of Computer Sci. and Eng., University of North Texas, USA
E-mail: lt0264@unt.edu
A. K. Bapatla

Dept. of Computer Sci. and Eng., University of North Texas, USA E-mail: anandkumarbapatla@my.unt.edu

V. K. V. V. Bathalapalli

Dept. of Computer Sci. and Eng., University of North Texas, USA

E-mail: vb0194@unt.edu

S. P. Mohanty

Dept. of Computer Sci. and Eng., University of North Texas, USA

E-mail: saraju.mohanty @unt.edu

E. Kougianos

Dept. of Electrical Engineering, University of North Texas, USA E-mail: elias.kougianos@unt.edu.

C. Ray

Dept. of Civil and Environmental Engineering, University of Nebraska-Lincoln, USA

E-mail: elias.kougianos @unt.edu.

we present Agriculture 4.0 and its applications, technology trends, available datasets, networking, and implementation challenges. We concentrate on Artificial Intelligence (AI) and Machine Learning (ML) technologies that support automation, as well as the Distributed Ledger Technology (DLT) which provides data integrity and security. Following an in-depth investigation of several architectures, we also provide a framework for smart agriculture that relies on the location of data processing. Open research problems of smart agriculture have been discussed from two perspectives - technology and communications. AI, ML, DLT, and Physical Unclonable Function (PUF) based hardware security fall under the technology group, whereas any Internet-based attacks, fake data injection and similar threats fall under the network research problem group. The survey aims to provide an in-depth study on recent works, challenges, and open research problems of smart agriculture to the researchers in this domain.

### **1** Introduction

By the end of the  $21^{st}$  century, world population is expected to reach 11 billion [9] and food consumption will surge at an unprecedented rate. Though a "Zero Hunger" goal by 2030 has been set, we are lagging behind the target [10]. It is estimated that about 800 million people are starving across the globe as of now [64]. Population increase makes the situation worse. To feed the world, food production is required to increase to 170% by 2050 [11].

A number of other factors are aggravating this situation: Firstly, urbanization is changing our diet. People now eat more animal protein than before. Annual consumption of animal protein per person from 1997 to 1999 was 36.4 kg and is expected to increase 20% by 2030. Secondly, natural resources are exhausted. Agricultural lands are becoming unsuitable for farming. Twenty-datasets, challenges, and open research problems. This five percent of the current arable land is not at all suitable and 44% is reasonably not suitable as of now. Due to water scarcity, 40% of the arable land is turning to unproductive land. Thirdly, urban expansion and deforestation for new arable lands also lead to rapid depletion of groundwater. Over-cultivation has also led to shorter fallow periods and reduced crop rotation whereas overgrazing leads to soil erosion. Food waste is another issue. 33% - 50% of the food is wasted worldwide. Finally, climate change is progressing very fast. It started affecting every aspect of food production. Over the last 50 years, greenhouse gas emissions have doubled with unpredictable rainfall and more instances of droughts and floods.

To combat these problems, the food and agro industries embrace "Agriculture 4.0", a smart and green movement, centered on science and technology [148]. Traditional farming is changing to viable, intelligent, efficient and environmentally friendly farming. New terms such as "smart farming," "digital farming," and "precision farming" are emerging. "Smart farming" is the same as "Smart Agriculture". Smart farming focuses on accessing data and using it to optimize complex systems and improve crop quality and yields. It also reduces human labor whereas precision farming targets optimization, precision, and crop specific solutions. The combination of the two is "digital farming." Fig. 1 shows the advantages of smart agriculture over traditional agriculture.



Fig. 1 Smart Agriculture Benefits Over Traditional Agriculture.

The majority of the survey works on smart agriculture address a specific area of research, e.g. either cybersecurity or AI. However, an overall idea of the research domain that provides a complete picture of smart agriculture is necessary for driving research in this important area of high social impact. In this survey

we discuss "Agriculture 4.0" a.k.a. "smart agriculture" and its applications, trends, technologies, networking, survey will provide a holistic information perspective on smart agriculture to researchers.

The rest of this survey is organized into eight sections. Internet-of-Agro-Things (IoAT) based Agriculture Cyber Physical Systems (A-CPS) are discussed in Section 2. Section 3 presents the smart agriculture architecture. Various applications of smart agriculture are described in Section 4. Challenges in implementing smart agriculture are presented in Section 5. Section 6 describes different technologies adapted in smart agriculture, whereas Section 7 discusses available datasets in the agricultural industry. Section 8 talks about the open research problems for the future and finally Section 9 concludes the paper.

# 2 IoAT based A-CPS

The Internet of Things (IoT) refers to physical things and devices with unique id connecting and sharing data through the Internet. Incorporating IoT in physical systems creates cyber physical systems (CPS). CPS is a mixed system of hardware and software. Defining any problem through CPS makes the process robust, seamless, and risk free and solves the problem optimally. The idea of CPS can be applied to any industry. When it comes to the agricultural industry, problems are defined through agricultural cyber physical systems (A-CPS) [149, 151]. Any IoT based smart agricultural system comprises of the stages described in Table 1. Locations of the occurrence of various stages have changed before and after the tinyML era [241].

### **3 Smart Agriculture Architecture**

After careful consideration of various architectures [66, 67, 78, 112, 118, 156, 186, 235 in the literature, we present the architecture of smart agriculture with three main layers and two connecting layers. Fig. 2(a) shows this architecture. Here, layers are defined according to the location (proximity to the occurrence) and their connections.

#### 3.1 Layer-1

Agriculture Device Layer is the first layer or the bottom most layer. It consists of sensors, animal paddocks, greenhouse, UAV, agro robots, automated tractors [5,

Stages	Descriptions	Before TinyML Era	TinyML Era
Data Collection	Things collect data.	Sensor level.	Sensor level.
Data Processing	Makes the collected data usable to the model.	Edge level	Edge level
Prognostic	Analyze the processed data using preset rules.	Cloud/Edge level.	Edge level
Solution	Solution of the issue is suggested.	Cloud/Edge level.	Edge level
Measures Taken	As per the solution, measures are taken.	IoT device level.	IoT device level.

Table 1 A-CPS Workflow Information.

6]. Data is collected and sensed in this layer. The devices, also known as distributed source nodes, monitor physical parameters and collect data in real time and transmit it to the gateway node at the next tier via the connectivity layer, thereby creating a Wireless Sensor Network (WSN). Fig. 2(b) depicts the information gathered by numerous sensors and cameras in various smart agriculture applications. For example, underground sensors and cameras on UAVs can gather data in a rice field and deliver it to the edge for analysis.

### 3.2 Layer-1a

This layer is in between layer-1 and layer-2. It is called Connectivity Layer-1. Layer-1 sends data to layer-2 through this layer. Different networks are used for different coverage areas as in Fig. 2(c). Depending on the application, various long and short range networks are chosen. For example, near range ZigBee, Wi-Fi, Z-Wave, Bluetooth, RFID, and NFC are used to send data from agriculture devices to the edge computing layer, while SigFox, LoRaWan, and NB-IoT are utilized for longer ranges [71]. Z-Wave is a good solution for a small farm in a distant community with restricted network connectivity. LoRaWan's low energy use and long distance transmission suit larger farms. Bluetooth low power is utilized for monitoring soil, air, and water management systems [84] and ZigBee for irrigation [131]. RFID technology is employed in smart agriculture for various purposes [80, 116, 175, 189, 208, 233]. The work in [255] used LoRa for water management purposes.

# 3.3 Layer-2

Edge Computing Layer is the second layer from the bottom. It consists of edge nodes. Data processing and encryption are performed in this layer. Because of the resource constraints at the edge layer, the prognostic and solution components were previously handled in the next layer. However, trained ML models can now compute and predict at this layer because of current hardware and AI edge developments. Both prognosis and inference can be done in the next layer if the work is resource intensive or not time sensitive. The edge computing layer, for example, performs the appropriate measures and notifies the farmer if a cow leaves its supposed territory in a livestock farm or needs milking.

For edge-AI initiatives, prognosis and solutions are also performed here. Various hardware boards are used for processing the data [71] at this layer e.g., Arduino UNO in [201] for a greenhouse application, Raspberry Pi in [136] for hydroponic systems, ESP8266 in [113] for connecting smart agricultural components, ESP32 in [37] for smart irrigation, Intel Edison in [34] for vertical agro warehouses, and BeagleBone in [18] for agrochemical process monitoring.

#### 3.4 Layer-2a

This the second connectivity layer. It is called Connectivity Layer-2. The edge computing layer sends data to layer-3 through this layer. Different networks are used for data transmission depending on the application needs. Fig. 2(c) lists the networks used in this layer. Cellular technologies including GPRS, LTE, 3G/4G, and 5G are used to send processed data from edge computing to the cloud. 5G features low latency, high dependability, wide coverage, fast data rate, and new frequency bands [235]. These features may boost communications capability of the smart agriculture. In [128], GPRS has been used for irrigation. New 5G initiatives have been presented in [15,65]. 6G cellular technology is the 5G network's successor. It's faster than current mobile networks. Smart agriculture will benefit from flexible decentralized models in edge computing, AI, and blockchain. Cloud management, framework, and integration with applications have been discussed for smart agriculture in [202]. A general smart cloud based system has been presented to facilitate smart farm remote sensing [104]. UAVs are being used as flying edge computing platforms to provide connectivity as and when needed [224]. Federated learning based approaches are



(a) Multi Layer Architecture of Smart Agriculture.



(b) Sensor Parameters measured in Agriculture Device Layer of Smart Agriculture Architecture.



(c) Networks used in Connectivity Layer of Smart Agriculture Architecture.

Fig. 2 Smart Agriculture Architecture Details.

also being used to address security and data privacy violation issue [119].

#### 3.5 Layer-3

This is the topmost layer. It is called *Cloud Computing Layer*. It stores data for future use. This layer is accessible through the Internet [78]. It was the computing and decision taking layer until recently [112, 128, 156], specially before tinyML. The cloud's high computational capacity enables it to complete a variety of complex jobs in an acceptable amount of time. However, cloud computing's limitations necessitate the emergence of new computing paradigms. Latency, high Internet bandwidth needs, data security and privacy are some of the limiting problems that impede the timesensitive monitoring and management of intelligent agriculture.

### 4 Smart Agriculture Applications

In this section, application areas of smart agriculture are discussed. Fig. 3 shows some application areas of smart agriculture.

#### 4.1 Crop Management

Analysis of economic, ecological, and social factors that go into crop selection, growing, and marketing is referred to as "crop management."

Cropping patterns are influenced by elements such as crop growth, water availability, labor, insurance, and environmental conditions. Changes in cropping patterns are influenced by environmental conditions. Traditional crop like rice, which requires a lot of water, cannot be grown in locations where water supplies are drying up and groundwater tables are decreasing. Different countries' export and import rules, as well as the agricultural product's market, all have an impact on crop selection. Once a crop has been chosen, the following step is to cultivate the crop. The IoT provides farmers with the most up-to-date technologies and sensors in the field to monitor plant growth. As an example, pests and insects that harm plant growth can be detected using ultrasonic sensors placed in the field. High-frequency sound waves are generated to eradicate pests after they have been detected, and the farmer is advised of their presence for additional assistance [238].

# 4.2 Soil Monitoring

Farming relies heavily on the availability of soil moisture. When a plant is growing it uses photosynthesis, respiration, transpiration, and mineral transport as a means of transporting nutrients. Farm decision-making relies heavily on soil monitoring. There are many elements that influence cropping patterns, including water availability and soil salinity.



Fig. 3 Applications of Smart Agriculture - Crop Management, Pest Control, Smart Irrigation, Livestock Monitoring Smart Greenhouse, UAV [1].

Soil health can be evaluated with the assistance of these variables. Field sensors collect soil temperature and humidity data, which is then uploaded to the cloud for further analysis. Based on the salinity content, soil nutrient level and soil humidity and temperature, cropping patterns are studied and determined. Water is required for photosynthesis, temperature regulation, and transport of food and nutrients for plant growth, hence soil moisture is an important consideration in the growth process.

For optimal plant growth, humidity governs the delivery of nutrients and the rate of transpiration. The optimal humidity for growing vegetables is between 50% and 60% [251]. A soil moisture sensor embedded in a plant's root analyzes soil moisture levels to ensure that water resources are optimally utilized [130, 182]. Soil conductivity has been measured along with soil moisture at low cost using RF propagation in available Wi-Fi bands [57]

# 4.3 Smart Irrigation

The use of cutting-edge irrigation technologies to improve crop quality and quantity is known as "smart irrigation." It saves water by watering plants in the most efficient way possible. There are two types of irrigation systems: (1) weather-based and (2) soil moisture sensorbased. An irrigation controller receives data from a local weather station and adjusts the amount of water. Sensors embedded in trees/grass correctly measure soil moisture. This sort of irrigation necessitates the use of precise humidity and air temperature measurements, as well as weather monitoring and knowledge of the field's cropping pattern. Sprinklers, for example, are actuated when data is transferred to the cloud [191]. The irrigation schedule for each farm area is determined by the sensor readings from the soil moisture probes. Optimal crop growth and 100% water efficiency can be achieved by precise irrigation scheduling and efficient actuation [158]. The irrigation system may be controlled by farmers using a smartphone app. Temperature, humidity, soil moisture and ultrasonic sensors in the field all feed into this irrigation system [163]. Farmers may activate the irrigation pumps to water their farms using a user-friendly mobile app on their smartphones, which connects to the cloud for analysis and control.

# 4.4 Livestock Monitoring

An important component of smart agriculture is livestock management. Farmers can monitor herd health, track grazing animals, and optimize breeding methods with an IoT-enabled livestock health monitoring system. Wearable collars or RFID tags can be used to measure vital signs like heart rate, blood pressure, and respiration rate in cattle. This serves two purposes: it saves labor while also treating animals in a timely manner, preventing the sickness from spreading further. GPS tracking is used for this purpose in [191]. It also serves to prevent any type of accident from occurring. Using RFID tags for animal identification and tracking is also common practice [240].

#### 4.5 Remote Sensing

Farmers can use remote sensing to acquire real-time agricultural data utilizing drones to map farm lands. Using crop health and agricultural information, they can also check crop yield. Remote sensing can map soil conditions and help farmers choose the best soil for a crop. Weeds and pests can be identified and controlled. Remote sensing's most essential use is weather forecasting. It can track rainfall, drought conditions, and water resources, notifying farmers on water availability and weather capital and crop planning can be done in advance [207]. When it comes to predicting production and plant growth, one of the most crucial metrics in crop cultivation, the Normalized Difference Vegetation Index (NVDI), has been used in [220]. Abiotic stress factors are monitored with the finest spatial resolution possible using remote sensing sensors on the farm field [95].

### 4.6 Smart Greenhouse

In the wake of global climate change and dwindling natural resources, the agricultural business welcomes farming approaches assisted by technology. The smart greenhouse is among them. It is an indoor environment specifically designed for the plants. It is a self-contained environment for agricultural monitoring that integrates IoT and AI/ML technologies. It safeguards the farm against wind, storm, and flooding. It boosts production without requiring manual involvement. Inside the greenhouse, solar-powered IoT-sensors are installed to monitor the health of the vegetables, fruits, and other horticultural crops. Using soil moisture sensors installed within the tree's root system, automatic drip irrigation can be carried out. If a predetermined threshold is met, the in-field actuators will water the farm accordingly. LED lighting can better serve the demands of plants. A regulated illumination of a certain wavelength and intensity can improve plant development and yield throughout the year.

Utilizing drip fertigation techniques, suitable amounts of minerals such as potassium, phosphorus, and others are applied to plants for optimal growth and health. As more technologies become available to farmers and the demand for organic fruits and vegetables grown with smart green techniques increases, greenhouse horticulture is becoming smarter [117]. In [223], a decision support-based IoT-friendly smart greenhouse system for enhancing rose plant productivity has been demonstrated.

### 4.7 Unmanned Aerial Vehicles

Unmanned Aerial Vehicles (UAVs), a.k.a. drones, are becoming increasingly prevalent in the contemporary agricultural economy. They are used for mapping crops, monitoring fields, remote sensing, fertigation, and weed detection. Drones can be a lifesaver when photographing vast agricultural tracts, mountainous regions, or remote locales.

To assess crop health, the *Normalized Difference Vegetation Index* (NVDI) is computed using photos captured by a drone. It determines the plant's water level, stress level, nutrient status, and pest infestation. It may steer the entire cultivation process of crops [72,155,183].

## 4.8 Autonomous Tractor

Innovative technologies are transforming the agricultural sector. Industrial IoT (IIoT) has advanced from crop management, soil monitoring, and smart irrigation to pest control, livestock management, and agrimarketing. In the near future, we can anticipate autonomous, intelligent, and smart farming equipment. An autonomous tractor is an integral component of these tools. This is a programmable autonomous vehicle. It can perform cultivation and fertilizer application. They are outfitted with GPS, lasers, and cameras and can operate autonomously without the need for farmer supervision.

Together with these smart tractors, an autonomous drone is employed for weed detection, pesticide spraying, field monitoring, and surveillance for sustainable agriculture [76]. The autonomous tractors used in orchards for spraying and mowing are equipped with a remote-assisted guide for conducting agricultural operations and perception systems for detecting impediments. The perception system employs cameras for obstacle detection and path identification based on geometry [153, 237].

### 4.9 Urban / Vertical Farming

In densely populated cities, the growing urbanization rate poses a serious problem. In these regions, a new way to farming has arisen to provide a sustainable farming option. Consequently, urban or vertical agriculture has gained popularity among urban residents. It occupies three-dimensional (3D) area for agriculture using controlled water, nutrients, minimum herbicides, and artificial lighting. The practical restriction of vertical farming systems is the generation of artificial light sources for plant development, as well as the associated high costs [83].

Hydroponics, as the name suggests, is a water-based system in which plants obtain all of their nutrition from a solution rich in elements. Continuous nutrient input is required in hydroponic systems. Using an app, a microcontroller can water the plants in a hydroponic system, as in in [234]. Aeroponics is similar to hydroponics but roots are misted instead of erupting in water. Aeroponic plants have been found to have a higher concentration of nutrients than hydroponic plants, according to research. This method is used to cultivate plants on board the International Space Station. Aquaponics, a relatively new agricultural method, is a hybrid of hydroponics and aquaculture in which the nutrients (e.g. phosphorus and nitrogen) are not added externally. Those nutrients are produced by the fish in the same tank.

### 4.10 Agriculture Marketing

An essential component of the expansion of a society's economy is the effective distribution of its goods and services. Inflation is caused by the presence of middle men, and as a result, both consumers and farmers suffer losses. Smart agriculture alters this dynamic. Using a variety of agro-marketing apps, farmers are now able to engage in direct consumer sales of their products.

A blockchain based on Ethereum has been utilized as a platform for the purpose of conducting trade talks between farmers and end consumers [187]. With the assistance of blockchain, a food supply chain has been established [41], beginning with the updating of a distributed ledger during the production phase and continuing all the way through to the distribution phase.

### 4.11 IoT Infrastructure

This area covers the IoT infrastructures e.g., various IoT sensors, cameras, connecting networks, efficient data collection, connectivity, low power and low cost devices. Efficient infrastructure increases the yield of a farm. An agricultural IoT platform has been presented in [232] for uninterrupted data collection though various agricultural things like sensors, cameras and UAVs. An asynchronous and reliable Sensor Network over White Spaces is presented in [193]. This bidirectional communication method sends different data to different sensor nodes at the same time and asynchronously receives the data from the nodes sent concurrently. In [188] a life cycle framework for energy efficient IoT in agriculture and its effects on various operational factors have been discussed.

# 5 Smart Agriculture Implementation Challenges

The processes of Smart Agriculture have modernized and facilitated traditional farming methods. However, developing a system for smart agriculture might not be as simple as putting few sensors out in the field to monitor conditions. Precision agriculture comes with a number of advantages, but it also poses a number of challenges for farmers and owners of agricultural businesses. These challenges need to be conquered if increased productivity and profits are to be realized as a result of precision agriculture. In this section, challenges of smart agriculture in modernizing the processes are discussed.

#### 5.1 Power Issues

The majority of "smart" agricultural activities involves the use of many machines, the operation of which requires significant quantities of power. It is not uncommon for farms to have quite high power requirements given their typically large land areas and the large number of required electronic components. This has been a significant barrier to the widespread adoption of automated procedures of this kind on large farms. Some of the proposed solutions include the use of clean energy derived from renewable sources such as solar, wind, and hydro, which would supply the machinery with continuous power that would not be interrupted [125]. Numerous researchers have shown an interest in this topic, and work is currently being done to adopt and improve the effectiveness of various renewable energy sources for smart farming [90, 184]. Inconsistent energy needs at various locations across the farm are just one of the challenges presented by these alternative power sources. Other energy or power related challenges include storing and transmitting the power that is generated. To solve these problems, an effective microgrid design must be developed. Research has been conducted in this field to develop an effective smart microgrid which can operate in conjunction with renewable energy sources [56, 61].

### 5.2 Power Consumption

IoT devices are connected through the Internet and generate huge amounts of data. This data is stored at the data centers which consume enormous amount of energy. Manufacturing of IoT devices like sensors, ICs, micro-controllers, and other semiconductor embedded devices also requires a lot of power. Additionally, some IoT devices are battery operated, hence they are required to consume low power. IoT devices are low power devices but transmission of data from machine to machine is one of the major sources of power consumption.

#### 5.3 Hardware Availability

A large number of sensors need to be interconnected for a successful IoT process. Once data is collected through "things" in the IoT network, they need to be processed and computed. Hence, hardware availability is another major challenge.

# 5.4 Hardware Security

The number of IoT devices is anticipated to be 50 billion by 2050 [55]. However, security of the hardware is compromised when low price is demanded. For IoT devices, two major security issues are Trojan and Side Channel Attacks (SCA). Hostile hardware modifications by the attacker are performed to manage the device secretly in the case of Trojan attacks. In [50], electron microscope scanning is performed on chips for detecting any additional gates present after comparing with a reference picture. A power efficient device specific Physically Unclonable Function (PUF) is proposed for IoT friendly embedded devices [195]. In the case of SCA, side channel signals e.g., electromagnetic emanation, power profiling and timing analysis, are utilized to obtain secret information like cryptographic keys. In [219], a real life SCA has been presented. Various solutions on hardware-assisted security (HAS) have been proposed in [33, 43, 103].

### 5.5 Physical Security of IoT Devices

Physical security of the hardware is another important challenge. IoT devices are prone to various physical attacks as they are installed outside, sometimes without any surveillance. Signal jamming, replication attack, eavesdropping, and physical damage of the device are some of these attacks [249]. Recently, due to the advancement in edge computing devices, edge initiatives in smart agriculture are gaining attention. Various physical security and safety issues of the IoT devices are addressed and safety measures are proposed in [12, 27, 74, 164].

### 5.6 Networking and Communication

Machine-to-Machine (M2M) communication plays a significant role in IoT based smart agriculture. They work collectively towards a final task. Data is shared through varied networks and communication protocols e.g., Zig-Bee, Wi-Fi, LoRA, SigFox, LTE, GPRS, and 5G. For a large agricultural farm, it is not a viable option for setting up and maintaining such networks due to the large cost. Various alternative but efficient methods are being proposed in different papers [49, 192, 254]. Other initiatives like SIL-IOT [246] have also been proposed. It is an agriculture thing, made by integrating Solar Insecticidal Lamps (SIL) with WSN for seamless communication. In [98], a LoRa based image transmission system has been proposed which can send the images from field camera.

#### 5.7 Connectivity Issue

Unavailability of high bandwidth Internet connection in villages hampers the cloud-centric computation and halts the progress of smart agriculture. Mountains and vast forests obstruct the line-of-sight (LOS) GPS communication [8].

### 5.8 Data Security and Privacy

To maintain data security and privacy, data encryption is a good practice. But due to the simple design of IoAT devices, data encryption is not always a viable option as encryption is a resource intensive process. As a result, smart automated agricultural solutions are vulnerable to various attacks. Yield and quality of the production can be affected negatively.

### 5.9 Scalability and Reliability

The size of agricultural farms varies from small to large. Depending on the size of the farm, the number and types of sensors also vary. As a result, a vast heterogeneous data set is generated from each farm. Hence, scalability plays an important role in smart agricultural solution. Reliability is another important factor. High reliability can reduce the number of redundant devices hence decreases the cost involved.

# 5.10 Big Data Challenge

In smart agriculture, the sensor nodes or cameras collect enormous amounts of data that may be broken down into many different categories. The conventional methods of processing such a vast amount of data are insufficient, which is where big data analysis comes in. Big data provides the capability to investigate very large datasets. It mitigates food security issues [46], provides predictive analysis and real time decision making, and it introduces new business models [178, 242]. All of these benefits can be attributed to the fact that the efficiency of the end-to-end supply chain in smart agricultural systems is improved.

Support Vector Machines (SVM) and Artifical Neural Networks (ANN) have been applied in order to integrate a big data platform in order to ensure the safety of the milk production chain [110]. The big data workflow in a smart agriculture system is depicted in Fig. 4 [46, 242]. It begins with the gathering of data at a variety of sensor nodes and finishes with the use of a variety of methodologies for data analysis, which can include both conventional and big data analysis.



Fig. 4 Big Data Work Flow in the Context of Smart Agriculture.

### 5.11 Challenges of AI

For sustainable, efficient, and automated agriculture, AI is a relevant choice. However, certain factors present challenges to employing AI in agriculture. The communication gap between the AI community and farmers is one of the major challenges. Multidisciplinary research can reduce this gap. Absence of regulations and policies is another challenge in applying AI. As sensors collect data from various regions, questions regarding data privacy and security come up. Cloud based solutions are also prone to cyber attacks. For edge-AI solutions, processing and computation of data is done locally. It reduces the chance of various attacks and solves the data privacy and regulation problem.

### 5.12 Technical Malfunction

Technical malfunction can impact a system negatively. Damaged sensors can wrongly sense data and errors can occur during decision making. It eventually can cause huge losses. For example, if the sensors in a crop field are damaged by hail, they sense wrong moisture data. As a result, the irrigation system may not work properly.

#### 5.13 Lack of Initial Capital Investment

Establishing new technologies in the cropland needs huge initial capital investment. However, in developing countries, where farmers have very thin profit margins, large capital investment is not a viable option. It stalls the extensive use of technologies in agriculture.

# 5.14 Unavailability of Uniform Standards

There is no uniform standard across the globe for smart agriculture solutions. It makes the solutions complex and expensive. A uniform standard is needed worldwide [8].

These aforementioned challenges pose significant bottleneck in implementing IoAT systems at various levels. The majority of them are common to any IoT system. However, each IoT system is impacted by these challenges at different degrees. For example, for IoAT system connectivity, huge amount of data generation, and awareness of the farmers about the available state-ofthe-art technologies are major barriers in modernizing agriculture. Theses challenges must be conquered in order to assure increased efficiency and maximized profits.

#### 6 Technologies for Smart Agriculture

In 2021, when the world was struggling through the pandemic, a digital and sustainable ecosystem was welcomed by the world's industry sectors. *Industry 5.0* arrived with digital transformation. The relationship of "human" and "machine" is re-defined. *Industry 5.0* speeds up the advent of *Agriculture 5.0*. Two main technologies, (1) artificial intelligence (AI) or machine learning (ML) and (2) distributed ledger technology (DLT), along with other technologies mentioned in Fig. 5 will guide this transition from *Agriculture 4.0* to *Agriculture 5.0*.

#### 6.1 Artificial Intelligence and Machine Learning

Machine intelligence that resembles human intelligence is referred to as artificial intelligence (AI). Progress in AI/ML technologies has shown great potential in variety of industries and research domains e.g., marketing [213], computer vision [141], multimedia forensics [139, 142, 144], healthcare [99], social media fake





(b) AI Tools for Smart Agriculture [122].

Fig. 5 Various Technologies for Smart Agriculture.

image and video detection [140, 143], gaming [167, 209], autonomous cars, and farming.

When used in agricultural applications, AI increases the efficiency of agricultural systems. Fig. 5(b) shows various AI tools applied in agriculture. These tools from different publications are being used in various areas of agriculture, as discussed in Table 2.

Varied AI technologies are being proposed based on the location of the computation. Research is ongoing to create deep neural network models with higher accuracy and fewer training parameters for edge AI initiatives, where the AI model operates on the restricted resource embedded system itself [137]. MobileNet [89], SqueezeNet [93], and EfficientNet [218] are examples of networks that execute depth wise convolution, data down sampling, and uniform model scaling down, respectively. The DNN size is reduced via quantization [47,101,247,257] and pruning [23,77,81,86,152,248,250]. The right hardware is just as crucial as the right algorithms.

AI has the potential to revolutionize the way we think about agriculture, making it possible for farmers to accomplish more with less work while also offering a variety of other benefits. However, artificial intelligence is not a stand-alone technology. It can serve as a bridge between traditional farming and the next stage of agricultural innovation. It collects and evaluates large amounts of data on a digital platform, determines the most effective course of action, and even executes that course of action when integrated with other forms of technologies. It facilitates improved decision-making, reduces the cost, limits the use of fertilizers and pesticides, brings higher profits, improves harvest quality and yield, monitors weather and soil health, and provides faster and accurate solutions from smart irrigation to vertical farming.

### 6.2 Distributed Ledger Technology

Distributed ledger technology consists of distinct nodes that help record, share, and synchronize the data transactions in their respective electronic ledgers instead of using central storage servers. Some examples of this technology include blockchain, DAG, Holochain, Tempo, and Hyperledger Fabric. Some ledger technologies, limitations, and applications are discussed here.

### 6.2.1 Blockchain

One of the recent digital technologies which has proved disruptive in the field of finances [40, 159] eliminating the need for centralized authorities and managing digital assets is the blockchain. Even though starting as financial solution, the blockchain has shown potential use cases in many industries like Real-time Secure IoT systems [94, 204], Smart Governance applications [75, 87], Digital Asset Copyright technologies [245, 252], Smart Healthcare [26,30,181] including Smart Agriculture and many other industries. Blockchain characteristics like hash linked data structure, cryptography verification, consensus based transaction processing helps in removing centralized authorities and provides more data security in an untrusted P2P network. IoT is another technology which has been a driving factor for smart agriculture systems in which using the IoT to make the farms climate resistant and also helps in achieving predicted yield even in case of environmental parameter fluctuations [127]. IoT three layered architecture consists of sensing layer, edge layer and cloud layer. Edge devices which are responsible for collecting data from sensing layer and process them form Edge Data Centers (EDC). They form a P2P network of EDCs to communicate and collectively work to process the information [180]. This machine-to-machine communication happening between EDCs can be made more secure and robust to network attacks by implementing the blockchain as solution. The relevance between application of blockchain in smart agriculture can be seen clearly in Fig. 6. Different applications of blockchain in smart agriculture can be seen in Table 3.

\_

Table 2 AI Technologies of Smart Agricul	ture.
--	-------

Applications	Descriptions	Works
Crop Management	Yield prediction, crop growth, dam- age estimation, and food supply chain researches come under crop manage- ment research. Varied AI/ML meth- ods e.g., Support Vector Machine (SVM), Gaussian Naive Bayes, vari- ous artificial neural networks (ANN), regression models, and clustering have been used.	SVM has been used in [185] to count the number of coffee fruits in a plant branch, rice yield prediction in [214], and in [198] to detect green immature citrus fruit. Gaussian Naive Bayes has been used to estimate cherries in a branch [21]. To estimate biomass of the grassland [17], to predict wheat [169], corn and soybean [109], corn [226], rice in hilly area [97], cotton [138, 253], wheat [190], maize [205], tea [211], general crop [53] yields, crop yield from soil parameters [124], crop nutrition disorder detection [212], and effect of crops on soil salt and water content [54]. Clustering is used in [199] for tomato detection from UAM images. Crop growth monitoring is done in [120].
Soil Management	Smart agricultural systems include soil property management, such as soil moisture, temperature, and nutri- ent content. It has two advantages: it increases crop output while also con- serving soil resources [63]. However, the procedure is time-taking and ex- pensive. Hence, a variety of low-cost and self-contained machine learning algorithms have been presented in or- der to develop a dependable soil man- agement system [122]. Sensor data, satellite photos, and UAV images are commonly used as input to machine learning models. In predictive anal- ysis, ANN, SVM, and autoencoders have been used.	For soil suitability evaluation, ANN and Multi-Layer Per- ceptrons (MLP) have been applied [236]. ML models have been used to forecast phosphorus in soil [59]. To extract geo-parcels from high-resolution photos, Deep Neural Net- works (DNN) were used, while MLP was used to forecast phosphorous content. The water retention capacity of soil in Brazilian coastal areas has been predicted using a radial basis function neural network [42]. Soil moisture may also be predicted from UAV-taken pictures using Boosted Re- gression Trees (BRT) [24] and ANN [25]. SVM was used to estimate the health and condition of soil moisture sensors, as well as the stage of degradation using Naive Bayes classi- fication [96]. From satellite pictures, autoencoder and SVM have been used to predict soil salinity [115].
Smart Irrigation	Smart agriculture systems include water management as a key com- ponent. Climate change is causing changes in rainfall patterns around the planet. Evapotranspiration is an important factor to consider when evaluating water supplies. In smart water management, a variety of AI technologies have been used.	In a crop field, deep reinforcement learning was applied for smart water management [39]. The water required for green- house organic crops was calculated using a multiple linear re- gression method, and then water valves were automatically operated using a LoRa Point-to-Point (P2P) network [44]. In [161], a study in Dehradun, India was used to suggest an ANN method for predicting evapotranspiration. Daily evapotranspiration has been predicted using ANN and the Penman-Monteith equation [22]. In an Edge-Fog-Cloud en- vironment, a smart irrigation system based on Long Short- Term Memory (LSTM) and Gated Recurrent Unit (GRU) models has been presented [52]. Another LSTM based ir- rigation system has been presented in [108] for precision agriculture. A neuro-drip irrigation system's spatial water distribution was predicted using ANN in [85].
Pest/Disease Control	Disease, pest, and weed control are re- quired to get the highest yield from an agricultural field. An automated sys- tem can help save time and expense. AI approaches are being proposed in several papers from this standpoint.	In the recent decade, rule-based systems [28,29,132,173,196] were developed, which were followed by Fuzzy Logic based systems [177, 206, 222, 228]. Various ANN have been used to detect various pests in a crop field [126] or to detect diseases in different crops [69,79,107,210]. For example, a channel–spatial attention module, incorporated with a backbone CNN and a Region Proposal Network (RPN), has been utilized for detecting different pests in a crop field [126]. Apple leaf disease is identified in [100] using the GoogleNet Inception network and Rainbow concatenation and in [145] using Mask R-CNN. For identifying pests in a tea plant, an incremental back propagation network was used with Correlation-based Feature Selection (CFS). The pest Tessaratoma Papillosa was localized using the CNN-based object detection model YOLOv3, and pest incidence was predicted with 90% accuracy using LSTM analysis of environmental data [45].

Continued on next page

Table 2 – Continued from previous page		
Applications	Descriptions	Works
		Anthrax on apple surface in an apple orchard was also de- tected using the YOLOv3 and YOLOv3-Dense models [239]. Single Seed Descent (SSD) has been shown to have an ac- curacy of 84% in detecting pests and 86% in classifying pests [134]. Using k-means clustering and a correspondence filter, pest detection and recognition were achieved [68]. In crop disease identification, various CNN-based models have been employed in [146,168,225]. Google object detection API with customized convolutional neural network has been used to detect good and bad coffee beans [91]. Recently, in [147], YOLOv5 and YOLOv8 have been used to detect pear, rice and wheat diseases in real time. In [58], the authors explored different light-duty computing platforms in IoAT-edge AI for plant disease detection.
Weed Control	Weed has a negative impact on yield. Weed control is thus another crucial aspect of smart agriculture. Weeds can be a nuisance at times. It's dif- ficult to tell them apart from crops. Artificial intelligence (AI) was first used in weed management in the early 2000s.	Hebbian synaptic modification was used with ANN to differ- entiate weeds from crops [14], and the accuracy gained was acceptable given the hardware present at the time. In [172], YOLOv3 was employed for less expensive weed manage- ment. To detect weeds, researchers used Counter Propa- gation (CP)-ANN with multi-spectral pictures [171] and a combination of auto encoder and SVM with hyper spectral images [170]. SVM was used in detecting weeds in grassland cropping in [35]. [111] uses a semi-supervised method to de- tect weeds.
Livestock Manage- ment	Animal welfare and livestock produc- tion have both benefited from AI/ML strategies in livestock management [122].	Authors in [60] used bagging ensemble learning for cattle, de- cision tree and C4.5 algorithm for calf [176], and Gaussian Mixture Models for pigs well-being [135]. AI aids in the opti- mization of livestock production efficiency. The work in [51] employed an ANN with back propagation to predict cattle rumen fermentation patterns from milk fatty acids. Faces of pigs were identified with 97 percent accuracy using CNN in [82]. SVM has been used to discover and warn about prob- lems in egg production for commercial hens [154], to estimate cattle weight trajectories for evolution [20], and to predict beef cattle skeletal weight [19]. In a robotic cow farm, ANN with Bayesian Regularization were utilized to forecast qual- ity milk output and lower cow heat stress levels [73]. In [165], a fully connected neural network was utilized to forecast cow illnesses.
Alternative Farming	Greenhouse farming and hydroponics are examples of alternative farming. In those systems, machine learning and deep learning approaches are em- ployed to provide superior and more accurate manage with less manpower.	Fully linked ANN and Root Mean Square Error (RMSE) are used to anticipate greenhouse air temperature [48]. ANN has been used to increase yield and growth for tomato [62] and basil yield in the greenhouse [166], greenhouse gas emissions and energy usage of wheat [114], and watermelon yield [157]. The humidity and temperature of a solar powered green- house were predicted using a Recurrent Neural Network (RNN) with back propagation [88]. RNN-LSTM has been used for climate prediction in [105]. In hydroponic systems, ANN and Bayesian Networks have been employed to fore- cast the required job [136].

# 6.2.2 Directed Acyclic Graph (DAG)

A Distributed Ledger with a directed acyclic graph (DAG) is used in IOTA Tangle, where each point represents a single transaction, and the arrows are authorizations. Each individual transaction is directed toward the previous parent transactions and is responsible for approv-

ing child transactions directly or indirectly as in Fig. 7. The IOTA Tangle is immutable by its novel consensus mechanism. Every transaction has a small Proof of Work (PoW) activity for consensus, making it impossible for the hacker to attack. An algorithm called Random Walk Monte Carlo is used for authentication. The Tangle grows and moves forward to increase the

Table 3 Blockchain Technologies in Smart Agricultur
---

Applications Secure Real-time Data Sharing	<b>Description</b> Data security and privacy is one of the important issues in IoT systems, robust cryptography is not a fea- sible solution considering resource- constrained IoT devices. Injection of false data and Denial-of-Service (DoS) are most common security at- tacks on IoT systems. There is a need for robust mechanism to share data between P2P network formed by EDC's	Works Blockchains as secure data sharing systems in smart agricul- ture is proposed in [129,200,244,256]. A robust identity man- agement system is proposed in [244] which will make use of private blockchain and prevent DOS attacks. A combination of SDN and blockchain networks is proposed in [200] which provides minimal overhead and provide robust security in agriculture systems. Work in [256] proposed an ethereum platform based system and made use of Practical Byzantine Fault Tolerant (PBFT) consensus mechanism. Key manage- ment system to securely share data is proposed in [129] which provides a secure, reliable and scalable key-management sys- tem which can be adapted in smart agriculture with numer- ous number of sensing nodes. The work in [31] proposes a distributed ledger based crop monitoring system which is a scalable blockchain environment in real-time data sharing and equip participating entities with decision support tools.
Community Farming and Local Markets	Community farming helps in sharing key data like weather, crop disease or product demand which can help as means for farmer to understand which crop should be cultivated. Lo- cal markets not only make fresh pro- duce available to the consumer di- rectly but also remove middle-men in- volved in the supply chain in order to maximize the profits of farmers. Pro- viding these two can make farming a profitable venture and lead more in- dividuals to participate in irrigation thereby increasing yield and provide food security.	A blockchain framework has been proposed which is imple- mented using Ethereum platform and can help in adapting ethical supply chains into farming [174]. Thereby, realizing profits to the farmers.
Supply-Chain Trace- ability	Food supply chains is a major com- ponent of Agriculture-CPS systems which enables the food products avail- able to all the corners of the world and also leverages collective working of or- ganizations which are geographically placed apart. Due to involvement of multiple entities and complex interac- tions between them, supply chain has become more complex and can lead many issues like Difficulty in tracking and tracing, processing delays leading to degradation of produce, resolving conflicts, lack of consumer confidence and recalls which increases unneces- sary wastage of produce.	Hyper ledger Fabric based systems are proposed in [133,243] as a case study to check the blockchain based supply chains. RFID has been integrated with blockchain systems to provide a transparent supply chain solution in [221]. A smart contract based solution for solving Single Point of Failure (SPOF) in centralized systems like ERP is proposed in [106]. The work in [32] proposed a system which uses ethereum platform to build a transparent supply chain for organic foods to increase customer confidence while consuming. A method integrating EPIC and blockchains is proposed by [123].
Farm Insurance	Farming yield majorly depends on the environmental parameters and cli- matic changes. This can lead to finan- cial in-stability of the farmers which may drive awsy many people from practicing farming as livelihood. Farm insurance is a financial setup where farmers will pay a fixed premium to a company and will be compensated with insured amount in case of dam- age to the crop from adverse climatic changes. These farm insurance com- panies need to assess the damage on the crop by following certain indexes pertaining to the weather parameters, blockchain can help in securely assess such parameters and provide a holis- tic indexes for computing the com- pensations.	A fraud detection and avoidance system for farm insurances based on blockchain is proposed in [160]. A drought based in- surance payout system is proposed and implemented in [162]. Authors of [16] proposed a blockchain based solution which leverages smart contracts on ethereum and hyperledger fab- ric platforms to design a insurance service system.



Fig. 6 IoT Network and Blockchain in Smart Agriculture.



Fig. 7 IOTA Tangle Architecture

weight to reach the consensus, and the transactions with light weight are orphaned and are not involved in the new consensus. Tangle can be more advantageous than blockchain for evading high fees, energy, and double spending vulnerabilities [230]. It has builtin two-layer solutions with MAM and STREAMS for data security and authentication and works on the principle of cryptography. It supports the concept of zerovalued transactions and can work with off-chain storage to skirt higher data volumes on the tangle [230].

# 6.2.3 Limitations of Blockchain

Blockchain has clearly shown many applications which can benefit the different smart agricultural operations but still adapting such technology in large scale and resource constrained smart agriculture systems is difficult. Blockchains as such consume large amounts of power and need large computational tools which is a major problem in IoT based smart agriculture systems. Hence, blockchains should be modified either by the structure or mechanisms in order to make them more feasible, reliable and scalable solutions for smart agriculture. A lightweight consensus mechanism Proof-of-Authentication is proposed in [179] which is IoT friendly and consumes much lower power compared to other

consensus protocols used in leading blockchain platforms. Another problem in implementing blockchain technologies in smart agriculture is the amount of data generated from the IoT networks, as the farms generate large amounts of data points will be collected every minute and analyzed. There is large constraint on blockchains on the amount of data that can be included in each block. Hence, efficient distributed off-chain solutions are on the rise. One such application which makes use of off-chain storage is the Inter-Planetary File System (IPFS) along with ethereum platform and leveraging smart contracts is proposed in [229]. Access management is also a major issue, a method to provide multi-level data access policies which is scalable and can be adapted into smart agriculture systems can be seen in [36].

Taking advantage of blockchain features of the incentive layer, an application for saving resources while farming is developed. This layer defines the minimum amount of transaction fees needed to perform actions on the blockchain. However, the paper proposes a novel idea using the same transactions as incentives to the farmers in the form of insurance once the electricity and water units are equaled to a prior condition to minimize carbon footprint, thus enhancing smart agriculture features through robust security [231]. To improve the security features for integrity and authentication and avoid central and blockchain limitations, IOTA Tangle is implemented to send the smart agricultural data in a secured method [230].

### 7 Datasets for Smart Agriculture Research

Smart devices are used in precision farming to collect data and analyze the yearly crop yields, supply chain transactions, and livestock management. The data stored using modern techniques in smart agriculture is then utilized for further research to ensure the availability of resources for future generations. Table 4 gives different datasets of various formats that were used for study in the current survey paper.

### 7.1 Crop Yield and Production

Many different sensors are used for gathering data on crop conditions, acreage, and yearly yields of the land. The total amount of crop yield data can be estimated by the ratio of the amount of produce to the harvested area, which is in terms of tonnes per hectare. The annual reports that the USDA gives have information regarding the yield, production estimates, costs in agriculture, livestock, plants, and agriculture census. The

Table 4 Datasets for Smart Agricultur	e.
---------------------------------------	----

Dataset Format	Dataset Purpose	Source	Link
.php	Crop Yield & Production	USDA & NASS	https://www.nass.usda.gov/Charts_and_Maps/
.gis	Crop Condition & soil moisture	Crop-CASMA	https://nassgeo.csiss.gmu.edu/CropCASMA/
.jpg	PlantVillage	Mendeley Data	https://data.mendeley.com/datasets/ tywbtsjrjv/1
.mdb	Soil Health & characterization	NCSS	https://new.cloudvault.usda.gov/index.php/ s/7iknp275KdTKwCA
.php, .txt	Pesticide use in agriculture	USGS	https://water.usgs.gov/nawqa/pnsp/usage/ maps/
Tableau	Water use in Agriculture	USGS	https://labs.waterdata.usgs.gov/ visualizations/water-use-15
.jpeg	Groundwater nitrate contamina- tion	USGS	https://prd-wret.s3.us-west- 2.amazonaws.com/assets/palladium/ production/s3fs-public/thumbnails/image/ wss-nitrogen-map-us-risk-areas.jpg
.png, .pdf	Disaster analysis	USDA & NASS	https://www.nass.usda.gov/ Research_and_Science/Disaster-Analysis/

prices, labor values, and production values change monthly the images of healthy crops and crops infected with disand annually and are presented in [13]. the eases. These images are used for training and testing of

### 7.2 Crop Condition and Soil Moisture

The crop yield can depend on many different factors, one of which is soil moisture. With humidity sensing, water availability, predicting weather disasters, and planning yearly crops can be performed easily. For different stages of agriculture to be realized quickly, soil data plays a vital role for the farmers to act accordingly. The NASA and USDA-NASS collaboratively have developed an application that collects the data in Geographic information system mapping format (.gis) [227], called Crop-CASMA, which is a web-based geospatial application used to measure the moisture of the soil and cropping conditions. The data of the moisture level and the state of the crops is given in Fig. 8(a).

### 7.3 Plant Diseases

Infections and diseases can change the core genetics and nutritions present in the crops, which can be harmful to consumers. Each plant species has its unique pattern and condition. Specific datasets of different plant infections is available in [3,92]. Fig. 9(a) and Fig. 9(b) show the images of healthy crops and crops infected with diseases. These images are used for training and testing of applications for predicting crop sickness to improve the harvest yield and productivity. Sample pomegranate images of different grades and qualities from the Pomegranate Fruit Dataset are shown in Fig. 9(c) [7].

### 7.4 Soil Health and Characterization

Survey conducted to know soil characteristics explain the features and properties of the soil. Farmers, engineers, and land agents are the end gainers of the soil survey data.

The classification of the soil database is provided by the National Cooperative Soil Survey (NCSS) with a pedon number associated with each report. The pedon gives a three-dimensional structure of the soil and explains the components present inside the ground. The properties of the earth like phosphorous, sand, water content, rock fragments, salt, pH levels are collected from the primary data characterization. Each report can be downloaded directly from the site in the form of graphs and text format by giving key details of country, state, and county.



(a) Dated November 07, 2021



(b) Dated March 18, 2022

Fig. 8 Crop Condition and Soil Moisture in the United States [227].

### 7.5 Pesticide Use in Agriculture

For controlling weeds, insects, and fungi, many pesticides are used, but excessive usage can pollute groundwater and kill other microorganisms that are necessary for the health of the soil. The USGS creates a database in the form of tables, graphs, and maps annually for showing the amount of pesticide used in farming [217]. The map gives a more detailed view of how much of the pesticide is used on the agricultural land in pounds per square mile, and graphs illustrate the estimated usage of fertilizers in millions of pounds each year.

### 7.6 Water Use in Agriculture

For farming livestock and crops, water plays an important role. Water is available for agriculture from both surface and groundwater [215]. Rivers and lakes are the primary sources for forming surface water, whereas groundwater resides in the sand cracks, rocks, and soil. The data for water usage are collected and updated every five years by the USGS in terms of billion gallons per day. The data collected illustrates that most of the water usage is in agriculture and industry [216].



(a) Healthy Plant Leaves - Potato [3], Peach [3], and Chinese Cabbage [2] (From Left to Right)



(b) Infected Plant Leaves - Potato [3], Peach [3], and Chinese Cabbage [2] (From Left to Right)



(c) Pomegranates of Different Grades and Different Qualities. G represents Grade and Q represents Quality [7].

Fig. 9 Sample Images from Various Plants and Fruits Datasets

### 7.7 Groundwater Nitrate Contamination

Nitrate is an oxidized form of nitrogen that helps in growing plants and crops. It is a compound that can be found naturally in the soil, but this compound can decrease if the land is extensively farmed. In order to increase the nutrients in the earth, different artificial nitrogen chemicals are used for growing crops. These nitrogen fertilizers can be harmful when they go inside crops, livestock, water, and groundwater. The USGS has designed a model to analyze how much groundwater is contaminated through the nitrate [197].

#### 7.8 Disaster Analysis

As the global temperatures, changing landscapes, and uncertain risks increase, the agriculture sector is facing adverse effects and threats. The disasters should be analyzed and estimated before they occur because the farmer has to be prepared for planning the crops accordingly. USDA and NASS are conducting research studies for disaster analysis in real-time. Most of the datasets are collected using sensors and geospatial methods to study the time when the disaster arises [4]. Sentinel-1, a Synthetic Aperture Radar is a modern tool that is used for studying and monitoring floods [38].

### 8 Smart Agriculture Open Research Problems

The open research problems of Agriculture 4.0 and Agriculture 5.0 are discussed in this section. Depending on the research objective, we can divide them into two major subgroups.

### 8.1 Technology Perspective

As aforementioned in Section 5, smart agriculture faces a number of implementation challenges. These issues must be addressed by the adaptation of existing and future technologies. Most smart agricultural AI models were cloud-based, cloud-edge-based, or cloud-fog-edgebased up until now. The shift in computing paradigm has been aided by hardware advancement. According to [203], adding intelligence to IoT devices is the new trend. Network availability, latency, and bandwidth are no longer obstacles in the operation of a viable, uninterrupted agriculture system. This opens up a new path to researches. The use of edge AI in smart agriculture is a huge deal that will become increasingly popular in the coming years. Fig. 10(a) depicts a variety of open research challenges in the context of technology. The following fields of study offer a lot of promise:

- Exploring tinyML devices with low power and low latency and solar powered devices.
- Low computing methods for IoT devices.
- Extreme temperature operable sensors.
- Data compression techniques.
- Quantization and pruning methods for deep learning models.
- Research on unsupervised and semi-supervised learning.
- Real time data processing and computation.
- More public dataset generation.
- UAV taken images accessible by the public.
- Thermal and Infrared image dataset.
- Annotated dataset for image segmentation.

These aren't the only research areas available. Other topics to work with include blockchain-based data privacy and integrity, as well as service-based smart agriculture applications, hardware security include the following:

 Blockchain-enabled IoT applications that emphasize immutable data storage techniques.

- Optimizing computing capabilities, lowering design
- time, and increasing efficiency are on the list too.Research on PUF, a hardware signature [102, 121], is a significant area of study.
- Vulnerability of PUF to environmental factors such as rain, herbicides, fertilizers, and chemicals.
- Reliability and tamper resistance of PUF.

### 8.2 Network Perspective

Network connectivity of smart agriculture is a critical component that uses various ICT to connect remote equipment, devices, and facilitates data transfer. Various security vulnerabilities have arisen as a result of developing unprotected network layer protocols for restricted resource IoT devices. Fig. 10(b) shows a classification of research difficulties from network perspective needs to be addressed including the following:

- Research on alternative networks in case of natural disasters.
- Methods for real time data processing even if the network is congested.
- Robust and resource friendly data privacy and security techniques.
- Minimize the blind spots by expanding the coverage area of the network.
- Cost effective methods for easy maintenance of instruments.
- Preventive approaches to counteract physical damages of the devices by adversaries [249].
- Improved network routing algorithm to avoid various network attacks.
- Efficient encryption and hardware authentication algorithms.
- Making troubleshooting mechanisms simple.
- Affordable equipment.

# 9 Conclusions

We regard "Let food be thy medicine" more than ever in today's environment since good food strengthens our immunity. Agriculture, food security, and the food supply chain have all become more vital in recent years. This article provides a comprehensive overview of current smart agriculture research efforts. It covers everything from the latest technological advances to open research concerns in this field. The authors anticipate that this paper will provide a broad overview of smart agriculture technology, challenges, and research issues.

Traditional agriculture has been changed into a smart, intelligent, and automated agriculture as a result of



(b) Network and Communication Perspective

Fig. 10 Open Research Problems of Smart Agriculture.

technological breakthroughs and the rapid rise of ICT. By implementing sustainable, green farming, decreasing the use of pesticides and fertilizers, and maximizing the use of natural resources, smart agriculture decreases the carbon footprint. It will also address other issues such as climate change and diseases such as cancer.

Agriculture 5.0 [194] will arrive in the agricultural industry soon. This will increase production while also ensuring the system's long-term viability. The same trend will be seen in developing countries as in wealthy countries. Humanity will welcome the production and delivery of food in a way that is both economically and environmentally efficient like never before [70].

# **Compliance with Ethical Standards**

The authors declare that they have no conflict of interest and there was no human or animal testing or participation involved in this research. All data were obtained from public domain sources.

# Acknowledgments

An extended version of this paper has been archived in [150].

# References

1. image: Freepik.com

- Chinese Cabbage Disease Dataset. https: //www.kaggle.com/giane901/chinese-cabbagedisease-detection
- 3. Data for: Identification of Plant Leaf Diseases Using a 9layer Deep Convolutional Neural Network. Accessed on 21 October 2021
- Disaster Analysis. https://www.nass.usda.gov/ Research\_and\_Science/Disaster-Analysis/. Accessed on 30 October 2021
- 5. Getting Started with Drones in Agriculture. https://extensionpublications.unl.edu/assets/ html/g2296/build/g2296.htm. Accessed on 20 September 2021
- 6. Livestock Sensors. https://www.precisionag.com/ in-field-technologies/sensors/using-iot-toincrease-efficiency-productivity-for-livestock/. Accessed on 20 September 2021
- Pomegranate Fruit Dataset. https://www.kaggle.com/ kumararun37/pomegranate-fruit-dataset
- Precision Agriculture Challenges. https: //teks.co.in/site/blog/precision-agriculturetop-15-challenges-and-issues/
- UN Report. https://www.un.org/devFelopment/desa/ en/news/population/world-population-prospects-2019.html. Accessed on 10 September 2021
- 10. UN Report: Food. https://www.un.org/en/globalissues/food
- 11. World Government Summit. https:// www.worldgovernmentsummit.org/api/publications/ document?id=95df8ac4-e97c-6578-b2f8ff0000a7ddb6. Accessed on 10 September 2021
- Afman, J., Ciarletta, L., Feron, E., Franklin, J., Gurriet, T., Johnson, E.N.: Towards a New Paradigm of UAV Safety. CoRR abs/1803.09026 (2018). URL http: //arxiv.org/abs/1803.09026
- of Agriculture, U.S.D., NASS: Census of Agriculture. https://www.nass.usda.gov/Charts\_and\_Maps/ (2021). Accessed on 30 October 2021
- Aitkenhead, M., Dalgetty, I., Mullins, C., McDonald, A.J.S., Strachan, N.J.C.: Weed and crop discrimination using image analysis and artificial intelligence methods. Computers and electronics in Agriculture **39**(3), 157–171 (2003)
- Alahmadi, A., Alwajeeh, T., Mohanan, V., Budiarto, R.: Wireless sensor network with always best connection for internet of farming. In: Powering the internet of things with 5G networks, pp. 176–201. IGI Global (2018)
- Aleksieva, V., Valchanov, H., Huliyan, A.: Smart contracts based on private and public blockchains for the purpose of insurance services. In: Proceedings International Conference Automatics and Informatics (ICAI). IEEE (2020). DOI 10.1109/icai50593.2020.9311371
- Ali, I., Cawkwell, F., Dwyer, E., Green, S.: Modeling managed grassland biomass estimation by using multitemporal remote sensing data—a machine learning approach. IEEE Journal of Selected Topics in Applied Earth Observations and Remote Sensing 10(7), 3254–3264 (2016)
- Ali, T.A., Choksi, V., Potdar, M.B.: Precision agriculture monitoring system using green internet of things (giot). In: Proceedings of 2nd International Conference on Trends in Electronics and Informatics (ICOEI), pp. 481– 487. IEEE (2018)
- Alonso, J., Castañón, Á.R., Bahamonde, A.: Support vector regression to predict carcass weight in beef cattle in advance of the slaughter. Computers and electronics in agriculture **91**, 116–120 (2013)

- Alonso, J., Villa, A., Bahamonde, A.: Improved estimation of bovine weight trajectories using support vector machine classification. Computers and electronics in agriculture 110, 36–41 (2015)
- Amatya, S., Karkee, M., Gongal, A., Zhang, Q., Whiting, M.D.: Detection of cherry tree branches with full foliage in planar architecture for automated sweet-cherry harvesting. Biosystems engineering 146, 3–15 (2016)
- Antonopoulos, V.Z., Antonopoulos, A.V.: Daily reference evapotranspiration estimates by artificial neural networks technique and empirical equations using limited input climate variables. Computers and Electronics in Agriculture 132, 86–96 (2017)
- Anwar, S., Sung, W.: Compact deep convolutional neural networks with coarse pruning. arXiv preprint arXiv:1610.09639 (2016)
- Araya, S.N., Fryjoff-Hung, A., Anderson, A., Viers, J.H., Ghezzehei, T.A.: Machine learning based soil moisture retrieval from unmanned aircraft system multispectral remote sensing. In: Proceedings of IEEE International Geoscience and Remote Sensing Symposium (IGARSS 2020), pp. 4598–4601 (2020). DOI 10.1109/ IGARSS39084.2020.9324117
- Arif, C., Mizoguchi, M., Setiawan, B.I., et al.: Estimation of soil moisture in paddy field using artificial neural networks. arXiv preprint arXiv:1303.1868 (2013)
- Azaria, A., Ekblaw, A., Vieira, T., Lippman, A.: MedRec: Using blockchain for medical data access and permission management. In: Proceedings 2nd International Conference on Open and Big Data (OBD). IEEE (2016). DOI 10.1109/obd.2016.11
- 27. Ballal, M.S.: Online electricity theft detection and prevention scheme for smart cities. IET Smart Cities 2, 155-164(9) (2020). URL https: //digital-library.theiet.org/content/journals/ 10.1049/iet-smc.2020.0045
- Balleda, K., Satyanvesh, D., Sampath, N., Varma, K., Baruah, P.: Agpest: An efficient rule-based expert system to prevent pest diseases of rice & wheat crops. In: Proceedings of IEEE 8th International Conference on Intelligent Systems and Control (ISCO), pp. 262–268. IEEE (2014)
- Banerjee, G., Sarkar, U., Ghosh, I.: A radial basis function network based classifier for detection of selected tea pests. International Journal of Advanced Research in Computer Science and Software Engineering 7(5), 665–669 (2017)
- Bapatla, A.K., Mohanty, S.P., Kougianos, E.: Pharma-Chain: A Blockchain to Ensure Counterfeit Free Pharmaceutical Supply Chain. CoRR abs/2202.02592 (2022). URL https://arxiv.org/abs/2202.02592
- Bapatla, A.K., Mohanty, S.P., Kougianos, E.: sFarm: A Distributed Ledger Based Remote Crop Monitoring System for Smart Farming. In: L.M. Camarinha-Matos, G. Heijenk, S. Katkoori, L. Strous (eds.) Internet of Things. Technology and Applications, pp. 13–31. Springer International Publishing, Cham (2022)
- 32. Basnayake, B.M.A.L., Rajapakse, C.: A blockchain-based decentralized system to ensure the transparency of organic food supply chain. In: Proceedings International Research Conference on Smart Computing and Systems Engineering (SCSE). IEEE (2019). DOI 10.23919/ scse.2019.8842690
- 33. Bathalapalli, V.K.V.V., Mohanty, S.P., Kougianos, E., Yanambaka, V.P., Baniya, B.K., Rout, B.: A puf-based approach for sustainable cybersecurity in smart agriculture. In: Proceedings of 19th OITS International Conference on Information Technology (OCIT), pp. 375–380 (2021). DOI 10.1109/OCIT53463.2021.00080

- 34. Bhowmick, S., Biswas, B., Biswas, M., Dey, A., Roy, S., Sarkar, S.K.: Application of iot-enabled smart agriculture in vertical farming. In: Advances in Communication, Devices and Networking, pp. 521–528. Springer (2019)
- Binch, A., Fox, C.: Controlled comparison of machine vision algorithms for rumex and urtica detection in grassland. Computers and Electronics in Agriculture 140, 123– 138 (2017)
- Biswas, S., Sharif, K., Li, F., Bairagi, A.K., Latif, Z., Mohanty, S.P.: GlobeChain: An interoperable blockchain for global sharing of healthcare data—a COVID-19 perspective. IEEE Consumer Electronics Magazine 10(5), 64–69 (2021). DOI 10.1109/mce.2021.3074688
- 37. Biswas, S.B., Iqbal, M.T.: Solar water pumping system control using a low cost esp32 microcontroller. In: Proceedings of IEEE Canadian conference on electrical & computer engineering (CCECE), pp. 1–5. IEEE (2018)
- Boryan, C.G., Yang, Z., Sandborn, A., Willis, P., Haack, B.: Operational Agricultural Flood Monitoring with Sentinel-1 Synthetic Aperture Radar. In: IGARSS 2018 -2018 IEEE International Geoscience and Remote Sensing Symposium, pp. 5831–5834 (2018). DOI 10.1109/ IGARSS.2018.8519458
- Bu, F., Wang, X.: A smart agriculture iot system based on deep reinforcement learning. Future Generation Computer Systems 99, 500–507 (2019)
- 40. Buterin, V.: A next-generation smart contract and decentralized application platform (2015)
- 41. Caro, M.P., Ali, M.S., Vecchio, M., Giaffreda, R.: Blockchain-based traceability in agri-food supply chain management: A practical implementation. In: Proceedings of IoT Vertical and Topical Summit on Agriculture - Tuscany (IOT Tuscany). IEEE (2018). DOI 10.1109/iot-tuscany.2018.8373021
- 42. Carvalho, G.N., Brandão, D., Haddad, D., do Forte, V., Ceddia, M.: A rbf neural network applied to predict soil field capacity and permanent wilting point at brazilian coast. In: Proceedings of International Joint Conference on Neural Networks (IJCNN), pp. 1–5 (2015). DOI 10.1109/IJCNN.2015.7280628
- Chakraborty, R.S., Bhunia, S.: Security against hardware trojan through a novel application of design obfuscation. In: Proceedings International Conference on Computer-Aided Design - ICCAD. ACM Press (2009). DOI 10.1145/ 1687399.1687424
- 44. Chang, Y.C., Huang, T.W., Huang, N.F.: A machine learning based smart irrigation system with lora p2p networks. In: Proceedings of 20th Asia-Pacific Network Operations and Management Symposium (APNOMS), pp. 1–4 (2019). DOI 10.23919/APNOMS.2019.8893034
- Chen, C.J., Huang, Y.Y., Li, Y.S., Chang, C.Y., Huang, Y.M.: An aiot based smart agricultural system for pests detection. IEEE Access 8, 180750–180761 (2020). DOI 10.1109/ACCESS.2020.3024891
- Chen, M., Mao, S., Liu, Y.: Big data: A survey. Mobile networks and applications 19(2), 171–209 (2014)
- Choi, Y., El-Khamy, M., Lee, J.: Towards the limit of network quantization. arXiv preprint arXiv:1612.01543 (2016)
- Codeluppi, G., Cilfone, A., Davoli, L., Ferrari, G.: AI at the Edge: a Smart Gateway for Greenhouse Air Temperature Forecasting. In: Proceedings of IEEE International Workshop on Metrology for Agriculture and Forestry (MetroAgriFor), pp. 348–353 (2020). DOI 10.1109/MetroAgriFor50201.2020.9277553

- 49. de la Concepcion, A.R., Stefanelli, R., Trinchero, D.: Adaptive wireless sensor networks for high-definition monitoring in sustainable agriculture. In: Proceedings IEEE Topical Conference on Wireless Sensors and Sensor Networks (WiSNet) (2014). DOI 10.1109/ wisnet.2014.6825511
- 50. Courbon, F., Loubet-Moundi, P., Fournier, J.J., Tria, A.: A high efficiency hardware trojan detection technique based on fast SEM imaging. In: Proceedings Design, Automation & Test in Europe Conference & Exhibition (DATE). IEEE Conference Publications (2015). DOI 10.7873/date.2015.1104
- Craninx, M., Fievez, V., Vlaeminck, B., De Baets, B.: Artificial neural network models of the rumen fermentation pattern in dairy cattle. Computers and electronics in agriculture 60(2), 226–238 (2008)
- Dahane, A., Benameur, R., Kechar, B., Benyamina, A.: An iot based smart farming system using machine learning. In: Proceedings of International Symposium on Networks, Computers and Communications (ISNCC), pp. 1–6 (2020). DOI 10.1109/ISNCC49221.2020.9297341
- Dahikar, S.S., Rode, S.V.: Agricultural crop yield prediction using artificial neural network approach. International journal of innovative research in electrical, electronics, instrumentation and control engineering 2(1), 683–686 (2014)
- Dai, X., Huo, Z., Wang, H.: Simulation for response of crop yield to soil moisture and salinity with artificial neural network. Field Crops Research 121(3), 441–449 (2011)
- 55. Dave, E., et al.: How the next evolution of the internet is changing everything. The Internet of Things (2011)
- Desai, A., Mukhopadhyay, I., Ray, A.: Techno-economicenvironment analysis of solar PV smart microgrid for sustainable rural electrification in agriculture community. In: Proceedings IEEE 48th Photovoltaic Specialists Conference (PVSC) (2021). DOI 10.1109/ pvsc43889.2021.9518454
- 57. Ding, J., Chandra, R.: Towards Low Cost Soil Sensing Using Wi-Fi. In: MobiCom 2019. ACM (2019). URL https: //www.microsoft.com/en-us/research/publication/ towards-low-cost-soil-sensing-using-wi-fi/
- Dockendorf, C., Mitra, A., Mohanty, S.P., Kougianos, E.: Lite-Agro: Exploring Light-Duty Computing Platforms for IoAT-Edge AI in Plant Disease Identification. In: IFIP International Internet of Things Conference. Springer (2023). DOI 10.1007/978-3-031-45882-8\_25
- Dong, W., Wu, T., Sun, Y., Luo, J.: Digital mapping of soil available phosphorus supported by ai technology for precision agriculture. In: Proceedings of 7th International Conference on Agro-geoinformatics (Agro-geoinformatics), pp. 1–5. IEEE (2018)
- Dutta, R., Smith, D., Rawnsley, R., Bishop-Hurley, G., Hills, J., Timms, G., Henry, D.: Dynamic cattle behavioural classification using supervised ensemble classifiers. Computers and electronics in agriculture **111**, 18–28 (2015)
- EBRAHIMI, M., KELATI, A., NKONOKI, E., KON-DORO, A., RWEGASIRA, D., DHAOU, I.B., TAAJA-MAA, V., TENHUNEN, H.: Creation of CERID: Challenge, education, research, innovation, and deployment "in the context of smart MicroGrid". In: Proceedings IST-Africa Week Conference (IST-Africa) (2019). DOI 10.23919/istafrica.2019.8764845
- 62. Ehret, D.L., Hill, B.D., Helmer, T., Edwards, D.R.: Neural network modeling of greenhouse tomato yield, growth and water use from automated crop monitoring data. Computers and electronics in agriculture **79**(1), 82–89 (2011)

- Eli-Chukwu, N.C.: Applications of artificial intelligence in agriculture: A review. Engineering, Technology & Applied Science Research 9(4), 4377–4383 (2019)
- 64. FAO, IFAD, UNICEF, WFP and WHO: Transforming food systems for affordable healthy diets (2020)
- 65. Faraci, G., Raciti, A., Rizzo, S., Schembra, G.: A 5g platform for unmanned aerial monitoring in rural areas: Design and performance issues. In: Proceedings of 4th IEEE Conference on Network Softwarization and Workshops (NetSoft), pp. 237–241. IEEE (2018)
- 66. Ferrández-Pastor, F.J., García-Chamizo, J.M., Nieto-Hidalgo, M., Mora-Martínez, J.: Precision agriculture design method using a distributed computing architecture on internet of things context. Sensors 18(6), 1731 (2018)
- 67. Ferrández-Pastor, F.J., García-Chamizo, J.M., Nieto-Hidalgo, M., Mora-Pascual, J., Mora-Martínez, J.: Developing ubiquitous sensor network platform using internet of things: Application in precision agriculture. Sensors 16(7), 1141 (2016)
- 68. Fina, F., Birch, P., Young, R., Obu, J., Faithpraise, B., Chatwin, C.: Automatic plant pest detection and recognition using k-means clustering algorithm and correspondence filters. International Journal of Advanced Biotechnology and Research 4(2), 189–199 (2013)
- Francl, L., Panigrahi, S.: Artificial neural network models of wheat leaf wetness. Agricultural and Forest Meteorology 88(1-4), 57–65 (1997)
- Fraser, E.D., Campbell, M.: Agriculture 5.0: reconciling production with planetary health. One Earth 1(3), 278– 280 (2019)
- Friha, O., Ferrag, M.A., Shu, L., Maglaras, L., Wang, X.: Internet of things for the future of smart agriculture: A comprehensive survey of emerging technologies. IEEE/CAA Journal of Automatica Sinica 8(4), 718–752 (2021). DOI 10.1109/JAS.2021.1003925
- Friha, O., Ferrag, M.A., Shu, L., Maglaras, L., Wang, X.: Internet of Things for the Future of Smart Agriculture: A Comprehensive Survey of Emerging Technologies. IEEE/CAA Journal of Automatica Sinica 8(4), 718–752 (2021). DOI 10.1109/jas.2021.1003925
- 73. Fuentes, S., Gonzalez Viejo, C., Cullen, B., Tongson, E., Chauhan, S.S., Dunshea, F.R.: Artificial intelligence applied to a robotic dairy farm to model milk productivity and quality based on cow data and daily environmental parameters. Sensors 20(10), 2975 (2020)
- Gao, Y., Foggo, B., Yu, N.: A Physically Inspired Data-Driven Model for Electricity Theft Detection With Smart Meter Data. IEEE Transactions on Industrial Informatics 15(9), 5076–5088 (2019). DOI 10.1109/TII.2019.2898171
- Geng, Q., Du, Y.: From blockchain to digital twin community: A technical framework for smart community governance. In: Proceedings International Conference on Public Management and Intelligent Society (PMIS) (2021). DOI 10.1109/pmis52742.2021.00068
- Gorli, R., Professor, A.: Future of Smart Farming with Internet of Things. Journal of Information technology and Its Applications 2 (2017)
- Guo, Y., Yao, A., Chen, Y.: Dynamic network surgery for efficient dnns. arXiv preprint arXiv:1608.04493 (2016)
- Gupta, M., Abdelsalam, M., Khorsandroo, S., Mittal, S.: Security and privacy in smart farming: Challenges and opportunities. IEEE Access 8, 34564–34584 (2020). DOI 10.1109/ACCESS.2020.2975142
- Hahn, F., Lopez, I., Hernandez, G.: Spectral detection and neural network discrimination of rhizopus stolonifer spores on red tomatoes. Biosystems Engineering 89(1), 93–99 (2004)

- Hamrita, T.K., Hoffacker, E.C.: Development of a "smart" wireless soil monitoring sensor prototype using rfid technology. Applied Engineering in Agriculture **21**(1), 139– 143 (2005)
- Han, S., Mao, H., Dally, W.: Compressing deep neural networks with pruning, trained quantization and huffman coding. arxiv 2015. arXiv preprint arXiv:1510.00149
- Hansen, M.F., Smith, M.L., Smith, L.N., Salter, M.G., Baxter, E.M., Farish, M., Grieve, B.: Towards on-farm pig face recognition using convolutional neural networks. Computers in Industry 98, 145–152 (2018)
- Haris, I., Fasching, A., Punzenberger, L., Grosu, R.: CPS/IoT ecosystem: Indoor vertical farming system. In: 2019 IEEE 23rd International Symposium on Consumer Technologies (ISCT). IEEE (2019). DOI 10.1109/ isce.2019.8900974
- 84. Hernández-Rojas, D.L., Fernández-Caramés, T.M., Fraga-Lamas, P., Escudero, C.J.: Design and practical evaluation of a family of lightweight protocols for heterogeneous sensing through ble beacons in iot telemetry applications. Sensors 18(1), 57 (2018)
- Hinnell, A., Lazarovitch, N., Furman, A., Poulton, M., Warrick, A.: Neuro-drip: estimation of subsurface wetting patterns for drip irrigation using neural networks. Irrigation science 28(6), 535–544 (2010)
- Hinton, G., Vinyals, O., Dean, J.: Distilling the knowledge in a neural network. arXiv preprint arXiv:1503.02531 (2015)
- Hjalmarsson, F.P., Hreioarsson, G.K., Hamdaqa, M., Hjalmtysson, G.: Blockchain-based e-voting system. In: Proceedings IEEE 11th International Conference on Cloud Computing (CLOUD) (2018). DOI 10.1109/ cloud.2018.00151
- Hongkang, W., Li, L., Yong, W., Fanjia, M., Haihua, W., Sigrimis, N.: Recurrent neural network model for prediction of microclimate in solar greenhouse. IFAC-PapersOnLine 51(17), 790–795 (2018)
- Howard, A.G., Zhu, M., Chen, B., Kalenichenko, D., Wang, W., Weyand, T., Andreetto, M., Adam, H.: Mobilenets: Efficient convolutional neural networks for mobile vision applications. CoRR abs/1704.04861 (2017). URL http://arxiv.org/abs/1704.04861
- Huang, K., Shu, L., Li, K., Yang, F., Han, G., Wang, X., Pearson, S.: Photovoltaic agricultural internet of things towards realizing the next generation of smart farming. IEEE Access 8, 76300–76312 (2020). DOI 10.1109/ access.2020.2988663
- 91. Huang, N.F.: Smart agriculture: real-time classification of green coffee beans by using a convolutional neural network. IET Smart Cities 2, 167-172(5) (2020). URL https://digital-library.theiet.org/ content/journals/10.1049/iet-smc.2020.0068
- 92. Hughes, D.P., Salathé, M.: An open access repository of images on plant health to enable the development of mobile disease diagnostics through machine learning and crowdsourcing. CoRR abs/1511.08060 (2015). URL http://arxiv.org/abs/1511.08060
- 93. Iandola, F.N., Han, S., Moskewicz, M.W., Ashraf, K., Dally, W.J., Keutzer, K.: Squeezenet: Alexnet-level accuracy with 50x fewer parameters and 0.5 mb model size. arXiv preprint arXiv:1602.07360 (2016)
- 94. Islam, M.A., Madria, S.: A permissioned blockchain based access control system for IOT. In: Proceedings IEEE International Conference on Blockchain (Blockchain) (2019). DOI 10.1109/blockchain.2019.00071
- 95. Jackson, T.: Soil water modeling and remote sensing. IEEE Transactions on Geoscience and Remote Sensing GE-24(1), 37–46 (1986). DOI 10.1109/tgrs.1986.289586

- 96. Jain, P., Choudhury, S.B., Bhatt, P., Sarangi, S., Pappula, S.: Maximising value of frugal soil moisture sensors for precision agriculture applications. In: Proceedings of IEEE / ITU International Conference on Artificial Intelligence for Good (AI4G), pp. 63–70 (2020). DOI 10.1109/AI4G50087.2020.9311008
- 97. Ji, B., Sun, Y., Yang, S., Wan, J.: Artificial neural networks for rice yield prediction in mountainous regions. The Journal of Agricultural Science 145(3), 249–261 (2007)
- Ji, M., Yoon, J., Choo, J., Jang, M., Smith, A.: LoRabased Visual Monitoring Scheme for Agriculture IoT. In: Proc. of IEEE Sensors Applications Symposium (SAS), pp. 1–6 (2019). DOI 10.1109/SAS.2019.8706100
- 99. Jiang, F., Jiang, Y., Zhi, H., Dong, Y., Li, H., Ma, S., Wang, Y., Dong, Q., Shen, H., Wang, Y.: Artificial intelligence in healthcare: past, present and future. Stroke and vascular neurology 2(4) (2017)
- 100. Jiang, P., Chen, Y., Liu, B., He, D., Liang, C.: Real-time detection of apple leaf diseases using deep learning approach based on improved convolutional neural networks. IEEE Access 7, 59069–59080 (2019)
- 101. Jin, Q., Yang, L., Liao, Z.: Adabits: Neural network quantization with adaptive bit-widths. In: Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, pp. 2146–2156 (2020)
- 102. Joshi, S., Mohanty, S.P., Kougianos, E.: Everything you wanted to know about PUFs. IEEE Potentials 36(6), 38– 46 (2017). DOI 10.1109/MPOT.2015.2490261
- 103. Ju, H., Jeon, Y., Kim, J.: A study on the hardwarebased security solutions for smart devices. In: Proceedings International Conference on Computational Science and Computational Intelligence (CSCI) (2015). DOI 10.1109/csci.2015.105
- 104. Junaid, M., Shaikh, A., Hassan, M.U., Alghamdi, A., Rajab, K., Reshan, A., Saleh, M., Alkinani, M.: Smart agriculture cloud using ai based techniques. Energies 14(16), 5129 (2021)
- 105. Jung, D.H., Kim, H.S., Jhin, C., Kim, H.J., Park, S.H.: Time-serial analysis of deep neural network models for prediction of climatic conditions inside a greenhouse. Computers and Electronics in Agriculture 173, 105402 (2020)
- 106. Kaid, D., Eljazzar, M.M.: Applying blockchain to automate installments payment between supply chain parties. In: Proceeding 14th International Computer Engineering Conference (ICENCO). IEEE (2018). DOI 10.1109/icenco.2018.8636131
- 107. Karmokar, B.C., Ullah, M.S., Siddiquee, M.K., Alam, K.M.R.: Tea leaf diseases recognition using neural network ensemble. International Journal of Computer Applications 114(17) (2015)
- Kashyap, P.K., Kumar, S., Jaiswal, A., Prasad, M., Gandomi, A.H.: Towards precision agriculture: IoT-enabled intelligent irrigation systems using deep learning neural network. IEEE Sensors Journal 21(16), 17479–17491 (2021)
- 109. Kaul, M., Hill, R.L., Walthall, C.: Artificial neural networks for corn and soybean yield prediction. Agricultural Systems 85(1), 1–18 (2005)
- 110. Kempenaar, C., Lokhorst, C., Bleumer, E., Veerkamp, R., Been, T., van Evert, F., Boogaardt, M., Ge, L., Wolfert, J., Verdouw, C., et al.: Big data analysis for smart farming: results of to2 project in theme food security. Tech. rep., Wageningen University & Research (2016)

- 111. Kethineni, K.K., Mitra, A., Mohanty, S.P., Kougianos, E.: WeedOut: An Autonomous Weed Sprayer in Smart Agriculture Framework using Semi-Supervised Non-CNN Annotation. In: IFIP International Internet of Things Conference. Springer (2023). DOI 10.1007/978-3-031-45878-1\_29
- 112. Khattab, A., Abdelgawad, A., Yelmarthi, K.: Design and implementation of a cloud-based iot scheme for precision agriculture. In: Proceedings of 28th International Conference on Microelectronics (ICM), pp. 201–204. IEEE (2016). DOI 10.1109/ICM.2016.7847850
- 113. Khoa, T.A., Man, M.M., Nguyen, T.Y., Nguyen, V., Nam, N.H.: Smart agriculture using iot multi-sensors: A novel watering management system. Journal of Sensor and Actuator Networks 8(3), 45 (2019)
- 114. Khoshnevisan, B., Rafiee, S., Omid, M., Yousefi, M., Movahedi, M.: Modeling of energy consumption and ghg (greenhouse gas) emissions in wheat production in esfahan province of iran using artificial neural networks. Energy 52, 333–338 (2013)
- 115. KLIBI, S., TOUŃSI, K., REBAH, Z.B., SOLAIMAN, B., FARAH, I.R.: Soil salinity prediction using a machine learning approach through hyperspectral satellite image. In: Proceedings of 5th International Conference on Advanced Technologies for Signal and Image Processing (ATSIP), pp. 1–6 (2020). DOI 10.1109/ ATSIP49331.2020.9231870
- Kodali, R.K., Jain, V., Karagwal, S.: Iot based smart greenhouse. In: Proceedings of IEEE region 10 humanitarian technology conference (R10-HTC), pp. 1–6. IEEE (2016)
- 117. Kodali, R.K., Jain, V., Karagwal, S.: IoT based smart greenhouse. In: Proceedings of IEEE Region 10 Humanitarian Technology Conference (R10-HTC), pp. 1–6 (2016). DOI 10.1109/R10-HTC.2016.7906846
- Köksal, Ö., Tekinerdogan, B.: Architecture design approach for iot-based farm management information systems. Precision Agriculture 20(5), 926–958 (2019)
- 119. Kumar, P., Gupta, G.P., Tripathi, R.: Pefl: Deep privacyencoding-based federated learning framework for smart agriculture. IEEE Micro 42(1), 33–40 (2022). DOI 10.1109/MM.2021.3112476
- 120. Kumar, S., Chowdhary, G., Udutalapally, V., Das, D., Mohanty, S.P.: GCrop: Internet-of-Leaf-Things (IoLT) for monitoring of the growth of crops in smart agriculture. In: Proceedings of IEEE International Symposium on Smart Electronic Systems (iSES)(Formerly iNiS), pp. 53– 56. IEEE (2019)
- Labrado, C., Thapliyal, H., Mohanty, S.P.: Fortifying vehicular security through low overhead physically unclonable functions. J. Emerg. Technol. Comput. Syst. 18(1) (2021). DOI 10.1145/3442443
- 122. Liakos, K.G., Busato, P., Moshou, D., Pearson, S., Bochtis, D.: Machine learning in agriculture: A review. Sensors 18(8), 2674 (2018)
- 123. Lin, Q., Wang, H., Pei, X., Wang, J.: Food safety traceability system based on blockchain and EPCIS. IEEE Access 7, 20698–20707 (2019). DOI 10.1109/ access.2019.2897792
- Liu, G., Yang, X., Li, M.: An artificial neural network model for crop yield responding to soil parameters. In: Proceedings of International Symposium on Neural Networks, pp. 1017–1021. Springer (2005)
   Liu, J., Chai, Y., Xiang, Y., Zhang, X., Gou, S., Liu,
- 125. Liu, J., Chai, Y., Xiang, Y., Zhang, X., Gou, S., Liu, Y.: Clean energy consumption of power systems towards smart agriculture: roadmap, bottlenecks and technologies. CSEE Journal of Power and Energy Systems 4(3), 273– 282 (2018). DOI 10.17775/cseejpes.2017.01290

- 126. Liu, L., Wang, R., Xie, C., Yang, P., Wang, F., Sudirman, S., Liu, W.: Pestnet: An end-to-end deep learning approach for large-scale multi-class pest detection and classification. IEEE Access 7, 45301–45312 (2019). DOI 10.1109/ACCESS.2019.2909522
- 127. Liu, Y., Ma, X., Shu, L., Hancke, G.P., Abu-Mahfouz, A.M.: From industry 4.0 to agriculture 4.0: Current status, enabling technologies, and research challenges. IEEE Transactions on Industrial Informatics 17(6), 4322–4334 (2021). DOI 10.1109/tii.2020.3003910
- López-Riquelme, J., Pavón-Pulido, N., Navarro-Hellín, H., Soto-Valles, F., Torres-Sánchez, R.: A software architecture based on fiware cloud for precision agriculture. Agricultural water management 183, 123–135 (2017)
- Ma, M., Shi, G., Li, F.: Privacy-oriented blockchain-based distributed key management architecture for hierarchical access control in the IoT scenario. IEEE Access 7, 34045– 34059 (2019). DOI 10.1109/access.2019.2904042
- Ma, Z., Pan, X.: Agricultural environment information collection system based on wireless sensor network. IEEE Global High Tech Congress on Electronics pp. 24–28 (2012)
- 131. Mafuta, M., Zennaro, M., Bagula, A., Ault, G., Gombachika, H., Chadza, T.: Successful deployment of a wireless sensor network for precision agriculture in malawi. International Journal of Distributed Sensor Networks 9(5), 150703 (2013)
- 132. Mahaman, B., Passam, H.C., Sideridis, A., Yialouris, C.: Diares-ipm: a diagnostic advisory rule-based expert system for integrated pest management in solanaceous crop systems. Agricultural Systems 76(3), 1119–1135 (2003)
- 133. Malik, S., Dedeoglu, V., Kanhere, S.S., Jurdak, R.: TrustChain: Trust management in blockchain and IoT supported supply chains. In: Proceedings IEEE International Conference on Blockchain (Blockchain). IEEE (2019). DOI 10.1109/blockchain.2019.00032
- Martin, V., Moisan, S.: Early pest detection in greenhouses. In: Proceedings of International Conference on Pattern Recognition (2008)
- 135. Matthews, S.G., Miller, A.L., PlÖtz, T., Kyriazakis, I.: Automated tracking to measure behavioural changes in pigs for health and welfare monitoring. Scientific Reports 7(1), 1–12 (2017)
- 136. Mehra, M., Saxena, S., Sankaranarayanan, S., Tom, R.J., Veeramanikandan, M.: Iot based hydroponics system using deep neural networks. Computers and electronics in agriculture 155, 473–486 (2018)
- 137. Merenda, M., Porcaro, C., Iero, D.: Edge machine learning for ai-enabled iot devices: A review. Sensors 20(9), 2533 (2020)
- Mitra, A., Beegum, S., Fleisher, D., Reddy, V.R., Sun, W., Ray, C., Timlin, D., Malakar, A.: Cotton yield prediction using random forest. arXiv preprint arXiv:2312.02299 (2023)
- 139. Mitra, A., Bigioi, D., Mohanty, S.P., Corcoran, P., Kougianos, E.: iFace 1.1: A Proof-of-Concept of a Facial Authentication Based Digital ID for Smart Cities. IEEE Access 10, 71791–71804 (2022). DOI 10.1109/ ACCESS.2022.3187686
- 140. Mitra, A., Mohanty, S.P., Corcoran, P., Kougianos, E.: A novel machine learning based method for deepfake video detection in social media. In: Proceedings of IEEE International Symposium on Smart Electronic Systems (iSES), pp. 91–96. IEEE (2020). DOI 10.1109/ iSES50453.2020.00031

- 141. Mitra, A., Mohanty, S.P., Corcoran, P., Kougianos, E.: Detection of Deep-Morphed Deepfake Images to Make Robust Automatic Facial Recognition Systems. In: Proceedings of 19th OITS International Conference on Information Technology (OCIT), pp. 149–154. IEEE (2021). DOI 10.1109/OCIT53463.2021.00039
- 142. Mitra, A., Mohanty, S.P., Corcoran, P., Kougianos, E.: iFace: A Deepfake Resilient Digital Identification Framework for Smart Cities. In: Proceedings of IEEE International Symposium on Smart Electronic Systems (iSES), pp. 361–366. IEEE (2021). DOI 10.1109/ iSES52644.2021.00090
- 143. Mitra, A., Mohanty, S.P., Corcoran, P., Kougianos, E.: A machine learning based approach for deepfake detection in social media through key video frame extraction. SN Computer Science 2(2), 98 (2021). DOI 10.1007/s42979-021-00495-x
- 144. Mitra, A., Mohanty, S.P., Corcoran, P., Kougianos, E.: EasyDeep: An IoT Friendly Robust Detection Method for GAN Generated Deepfake Images in Social Media. In: L.M. Camarinha-Matos, G. Heijenk, S. Katkoori, L. Strous (eds.) Internet of Things. Technology and Applications, pp. 217–236. Springer International Publishing, Cham (2022). DOI https://doi.org/10.1007/978-3-030-96466-5\_14
- 145. Mitra, A., Mohanty, S.P., Kougianos, E.: A Smart Agriculture Framework to Automatically Track the Spread of Plant Diseases Using Mask Region-Based Convolutional Neural Network. In: IFIP International Internet of Things Conference, pp. 68–85. Springer (2022). DOI 10.1007/978-3-031-18872-5\_5
- 146. Mitra, A., Mohanty, S.P., Kougianos, E.: aGROdet: a novel framework for plant disease detection and leaf damage estimation. In: IFIP International Internet of Things Conference, pp. 3–22. Springer (2022)
- 147. Mitra, A., Mohanty, S.P., Kougianos, E.: aGRodet 2.0: An Automated Real-Time Approach for Multiclass Plant Disease Detection. SN Computer Science 4(5), 657 (2023). DOI 10.1007/s42979-023-02076-6
- 148. Mitra, A., Mohanty, S.P., Kougianos, E.: Smart Agriculture – Demystified. In: IFIP International Internet of Things Conference. Springer (2023). DOI 10.1007/978-3-031-45878-1\_28
- 149. Mitra, A., Singhal, A., Mohanty, S.P., Kougianos, E.: eCrop: A Novel Framework for Automatic Crop Damage Estimation in Smart Agriculture. SN Computer Science 3(319) (2022). DOI 10.1007/s42979-022-01216-8
- 150. Mitra, A., Vangipuram, S.L.T., Bapatla, A.K., Bathalapalli, V.K.V.V., Mohanty, S.P., Kougianos, E., Ray, C.: Everything you wanted to know about smart agriculture. CoRR abs/2201.04754 (2022). URL https: //arxiv.org/abs/2201.04754
- Mohanty, S.P.: Internet-of-Agro-Things (IoAT) Makes Smart Agriculture. IEEE Consumer Electronics Magazine 10(4), 4–5 (2021)
- 152. Molchanov, P., Tyree, S., Karras, T., Aila, T., Kautz, J.: Pruning convolutional neural networks for resource efficient inference. arXiv preprint arXiv:1611.06440 (2016)
- 153. Moorehead, S.J., Wellington, C.K., Gilmore, B.J., Vallespi, C.: Automating orchards: A system of autonomous tractors for orchard maintenance. In: Proceedings of the IEEE international conference of intelligent robots and systems, workshop on agricultural robotics (2012)
- 154. Morales, I.R., Cebrián, D.R., Blanco, E.F., Sierra, A.P.: Early warning in egg production curves from commercial hens: A svm approach. Computers and electronics in agriculture **121**, 169–179 (2016)

- 155. Muniasamy, A.: Machine Learning for Smart Farming: AFocus on Desert Agriculture. In: 2020 International Conference on Computing and Information Technology (ICCIT-1441). IEEE (2020). DOI 10.1109/iccit-144147971.2020.9213759
- 156. Na, A., Isaac, W.: Developing a human-centric agricultural model in the iot environment. In: Proceedings of International Conference on Internet of Things and Applications (IOTA), pp. 292–297. IEEE (2016). DOI 10.1109/IOTA.2016.7562740
- 157. Nabavi-Pelesaraei, A., Abdi, R., Rafiee, S.: Neural network modeling of energy use and greenhouse gas emissions of watermelon production systems. Journal of the Saudi Society of Agricultural Sciences 15(1), 38–47 (2016)
- 158. Nagothu, S.K.: Weather based smart watering system using soil sensor and GSM. In: Proceedings of World Conference on Futuristic Trends in Research and Innovation for Social Welfare (Startup Conclave), pp. 1–3 (2016). DOI 10.1109/STARTUP.2016.7583991
- Nakamoto, S.: Bitcoin: A peer-to-peer electronic cash system. Cryptography Mailing list at https://metzdowd.com (2009)
- 160. Nath, I.: Data exchange platform to fight insurance fraud on blockchain. In: Proceedings IEEE 16th International Conference on Data Mining Workshops (ICDMW). IEEE (2016). DOI 10.1109/icdmw.2016.0121
- 161. Nema, M.K., Khare, D., Chandniha, S.K.: Application of artificial intelligence to estimate the reference evapotranspiration in sub-humid doon valley. Applied Water Science 7(7), 3903–3910 (2017)
- 162. Nguyen, T., Das, A., Tran, L.: NEO smart contract for drought-based insurance. In: Proceedings IEEE Canadian Conference of Electrical and Computer Engineering (CCECE). IEEE (2019). DOI 10.1109/ccece.2019.8861573
- 163. Ogidan, O.K., Afia, K.R.: Smart irrigation system with an android-based remote logging and control. In: Proceedings of IEEE AFRICON. IEEE (2019). DOI 10.1109/ africon46755.2019.9133953
- 164. Otuoze, A.O.: Electricity theft detection by sources of threats for smart city planning. IET Smart Cities 1, 52-60(8) (2019). URL https: //digital-library.theiet.org/content/journals/ 10.1049/iet-smc.2019.0045
- 165. P. S. Chatterjee, N.K.R., Mohanty, S.P.: LiveCare: An IoT based Healthcare Framework for Livestocks in Smart Agriculture. IEEE Transactions on Consumer Electronics (TCE) 67(4), 257–265 (2021)
- 166. Pahlavan, R., Omid, M., Akram, A.: Energy input-output analysis and application of artificial neural networks for predicting greenhouse basil production. Energy 37(1), 171–176 (2012)
- 167. Palaus, M., Marron, E.M., Viejo-Sobera, R., Redolar-Ripoll, D.: Neural basis of video gaming: A systematic review. Frontiers in human neuroscience **11**, 248 (2017)
- 168. Pallagani, V., Khandelwal, V., Chandra, B., Udutalapally, V., Das, D., Mohanty, S.P.: DCrop: A deep-learning based framework for accurate prediction of diseases of crops in smart agriculture. In: Proceedings of IEEE International Symposium on Smart Electronic Systems (iSES)(Formerly iNiS), pp. 29–33. IEEE (2019)
- 169. Pantazi, X.E., Moshou, D., Alexandridis, T., Whetton, R.L., Mouazen, A.M.: Wheat yield prediction using machine learning and advanced sensing techniques. Computers and electronics in agriculture **121**, 57–65 (2016)
- Pantazi, X.E., Moshou, D., Bravo, C.: Active learning system for weed species recognition based on hyperspectral sensing. Biosystems Engineering 146, 193–202 (2016)

- 171. Pantazi, X.E., Tamouridou, A.A., Alexandridis, T., Lagopodi, A.L., Kashefi, J., Moshou, D.: Evaluation of hierarchical self-organising maps for weed mapping using uas multispectral imagery. Computers and Electronics in Agriculture **139**, 224–230 (2017)
- 172. Partel, V., Kakarla, S.C., Ampatzidis, Y.: Development and evaluation of a low-cost and smart technology for precision weed management utilizing artificial intelligence. Computers and electronics in agriculture 157, 339–350 (2019)
- 173. Pasqual, G., Mansfield, J.: Development of a prototype expert system for identification and control of insect pests. Computers and Electronics in Agriculture 2(4), 263–276 (1988)
- 174. Paul, S., Joy, J.I., Sarker, S., Shakib, A.A.H., Ahmed, S., Das, A.K.: An unorthodox way of farming without intermediaries through blockchain. In: Proceedings International Conference on Sustainable Technologies for Industry 4.0 (STI). IEEE (2019). DOI 10.1109/ sti47673.2019.9068007
- 175. Peets, S., Gasparin, C., Blackburn, D., Godwin, R.: Rfid tags for identifying and verifying agrochemicals in food traceability systems. Precision Agriculture 10(5), 382– 394 (2009)
- 176. Pegorini, V., Zen Karam, L., Pitta, C.S.R., Cardoso, R., Da Silva, J.C.C., Kalinowski, H.J., Ribeiro, R., Bertotti, F.L., Assmann, T.S.: In vivo pattern classification of ingestive behavior in ruminants using fbg sensors and machine learning. Sensors 15(11), 28456–28471 (2015)
- 177. Peixoto, M.S., Barros, L.C., Bassanezi, R.C., Fernandes, O.A.: An approach via fuzzy systems for dynamics and control of the soybean aphid (2015)
- 178. Poppe, K., Wolfert, J., Verdouw, C., Renwick, A.: A european perspective on the economics of big data. Farm Policy Journal 12(1), 11–19 (2015)
- 179. Puthal, D., Mohanty, S.P.: Proof of authentication: IoTfriendly blockchains. IEEE Potentials 38(1), 26–29 (2019). DOI 10.1109/mpot.2018.2850541
- 180. Puthal, D., Mohanty, S.P., Nanda, P., Kougianos, E., Das, G.: Proof-of-authentication for scalable blockchain in resource-constrained distributed systems. In: Proceedings IEEE International Conference on Consumer Electronics (ICCE) (2019). DOI 10.1109/icce.2019.8662009
- 181. Rachakonda, L., Bapatla, A.K., Mohanty, S.P., Kougianos, E.: SaYoPillow: Blockchain-integrated privacy-assured IoMT framework for stress management considering sleeping habits. IEEE Transactions on Consumer Electronics 67(1), 20–29 (2021). DOI 10.1109/tce.2020.3043683
- 182. Radha, N., Swathika, R.: A Polyhouse: Plant Monitoring and Diseases Detection using CNN. In: Proceedings of International Conference on Artificial Intelligence and Smart Systems (ICAIS), pp. 966–971 (2021). DOI 10.1109/ICAIS50930.2021.9395847
- Radoglou-Grammatikis, P., Sarigiannidis, P., Lagkas, T., Moscholios, I.: A compilation of UAV applications for precision agriculture. Computer Networks 172, 107148 (2020). DOI 10.1016/j.comnet.2020.107148
- 184. Ram, S.K., Sahoo, S.R., Das, B.B., Mahapatra, K., Mohanty, S.P.: Eternal-thing: A secure aging-aware solarenergy harvester thing for sustainable IoT. IEEE Transactions on Sustainable Computing 6(2), 320–333 (2021). DOI 10.1109/tsusc.2020.2987616
- Ramos, P., Prieto, F.A., Montoya, E., Oliveros, C.E.: Automatic fruit count on coffee branches using computer vision. Computers and Electronics in Agriculture 137, 9–22 (2017)

- 186. Ray, P.P.: Internet of things for smart agriculture: Technologies, practices and future direction. Journal of Ambient Intelligence and Smart Environments 9(4), 395–420 (2017)
- 187. Revathy, S., Priya, S.S.: Blockchain based Producer-Consumer Model for Farmers. In: Proceedings of the 4th International Conference on Computer, Communication and Signal Processing (ICCCSP). IEEE (2020). DOI 10.1109/icccsp49186.2020.9315214
- 188. Ruan, J., Wang, Y., Chan, F.T.S., Hu, X., Zhao, M., Zhu, F., Shi, B., Shi, Y., Lin, F.: A Life Cycle Framework of Green IoT-Based Agriculture and Its Finance, Operation, and Management Issues. IEEE Communications Magazine 57(3), 90–96 (2019). DOI 10.1109/ MCOM.2019.1800332
- Ruiz-Garcia, L., Lunadei, L.: The role of rfid in agriculture: Applications, limitations and challenges. Computers and Electronics in Agriculture 79(1), 42–50 (2011)
- 190. Ruß, G., Kruse, R., Schneider, M., Wagner, P.: Data mining with neural networks for wheat yield prediction. In: Proceedings of Industrial Conference on Data Mining, pp. 47–56. Springer (2008)
- 191. S., K.: IoT in Agriculture : Smart Farming. International Journal of Scientific Research in Computer Science, Engineering and Information Technology pp. 181–184 (2018). DOI 10.32628/cseit183856
- 192. Sahitya, G., Balaji, N., Naidu, C.: Wireless sensor network for smart agriculture. In: Proceedings 2nd International Conference on Applied and Theoretical Computing and Communication Technology (iCATccT) (2016). DOI 10.1109/icatcct.2016.7912049
- 193. Saifullah, A., Rehman, M., Ismail, D., Lu, C., Liu, J., Chandra, R.: Enabling Reliable, Asynchronous, and Bidirectional Communication in Sensor Networks over White Spaces. In: ACM Conference on Embedded Networked Sensor Systems (SenSys'17). ACM (2017). URL https: //www.microsoft.com/en-us/research/publication/ enabling-reliable-asynchronous-bidirectionalcommunication-sensor-networks-white-spaces/
- 194. Saiz-Rubio, V., Rovira-Más, F.: From smart farming towards agriculture 5.0: A review on crop data management. Agronomy 10(2), 207 (2020)
- 195. Sankaran, S., Shivshankar, S., Nimmy, K.: LHPUF: Lightweight hybrid PUF for enhanced security in internet of things. In: Proceedings IEEE International Symposium on Smart Electronic Systems (iSES) (Formerly iNiS) (2018). DOI 10.1109/ises.2018.00066
- 196. Sarma, S.K., Singh, K.R., Singh, A.: An expert system for diagnosis of diseases in rice plant. International Journal of Artificial Intelligence 1(1), 26–31 (2010)
- 197. School, W.S.: Areas at high risk of nitrogen contamination of groundwater . https://www.usgs.gov/media/ images/areas-high-risk-nitrogen-contaminationgroundwater. Accessed on 30 October 2021
- 198. Sengupta, S., Lee, W.S.: Identification and determination of the number of immature green citrus fruit in a canopy under different ambient light conditions. Biosystems Engineering **117**, 51–61 (2014)
- 199. Senthilnath, J., Dokania, A., Kandukuri, M., Ramesh, K., Anand, G., Omkar, S.: Detection of tomatoes using spectral-spatial methods in remotely sensed rgb images captured by uav. Biosystems engineering 146, 16–32 (2016)
- 200. Sharma, P.K., Singh, S., Jeong, Y.S., Park, J.H.: Dist-BlockNet: A distributed blockchains-based secure SDN architecture for IoT networks. IEEE Communications Magazine 55(9), 78–85 (2017). DOI 10.1109/ mcom.2017.1700041

- 201. Shirsath, D., Kamble, P., Mane, R., Kolap, A., More, R.: Iot based smart greenhouse automation using arduino. International Journal of Innovative Research in Computer Science & Technology 5(2), 234–238 (2017)
- Shrawankar, U., Malik, L., Arora, S.: Cloud Computing Technologies for Smart Agriculture and Healthcare. Chapman and Hall/CRC (2021)
- 203. Signoretti, G., Silva, M., Andrade, P., Silva, I., Sisinni, E., Ferrari, P.: An Evolving TinyML Compression Algorithm for IoT Environments Based on Data Eccentricity. Sensors 21(12), 4153 (2021)
- 204. Singh, M., Singh, A., Kim, S.: Blockchain: A game changer for securing IoT data. In: Proceedings IEEE 4th World Forum on Internet of Things (WF-IoT) (2018). DOI 10.1109/wf-iot.2018.8355182
- 205. Singh, R.K., et al.: Artificial neural network methodology for modelling and forecasting maize crop yield. Agricultural Economics Research Review **21**(347-2016-16813), 5–10 (2008)
- 206. Siraj, F., Arbaiy, N.: Integrated pest management system using fuzzy expert system (2006)
- 207. Sishodia, R.P., Ray, R.L., Singh, S.K.: Applications of remote sensing in precision agriculture: A review. Remote Sensing 12(19), 3136 (2020). DOI 10.3390/rs12193136
- 208. Sjolander, A., Thomasson, J., Sui, R., Ge, Y.: Wireless tracking of cotton modules. part 2: Automatic machine identification and system testing. Computers and electronics in agriculture 75(1), 34–43 (2011)
- 209. Skinner, G., Walmsley, T.: Artificial intelligence and deep learning in video games a brief review. In: Proceedings of IEEE 4th International Conference on Computer and Communication Systems (ICCCS), pp. 404–408 (2019). DOI 10.1109/CCOMS.2019.8821783
- 210. Sladojevic, S., Arsenovic, M., Anderla, A., Culibrk, D., Stefanovic, D.: Deep neural networks based recognition of plant diseases by leaf image classification. Computational intelligence and neuroscience **2016** (2016)
- 211. Soheili-Fard, F., Salvatian, S.B.: Forecasting of tea yield based on energy inputs using artificial neural networks (a case study: Guilan province of iran). In: Proceedings of Biological Forum, vol. 7, p. 1432. Research Trend (2015)
- 212. Song, H., He, Y.: Crop nutrition diagnosis expert system based on artificial neural networks. In: Proceedings of Third International Conference on Information Technology and Applications (ICITA'05), vol. 1, pp. 357–362. IEEE (2005)
- 213. Soni, V.D.: Emerging roles of artificial intelligence in ecommerce. International Journal of trend in scientific research and development 4(5), 223–225 (2020)
- 214. Su, Y.x., Xu, H., Yan, L.j.: Support vector machine-based open crop model (sbocm): Case of rice production in china. Saudi journal of biological sciences 24(3), 537–547 (2017)
- 215. Survey, U.: Total Water Use in the United States. https://www.usgs.gov/special-topic/water-scienceschool/science/total-water-use-united-states (2015). Accessed on 30 October 2021
- 216. Survey, U.: Water use in the U.S. https: //labs.waterdata.usgs.gov/visualizations/wateruse-15/ (2015). Accessed on 30 October 2021
- 217. Survey, U.G.: Estimated Annual Agricultural Pesticide Use. https://water.usgs.gov/nawqa/pnsp/usage/maps/. Accessed on 30 October 2021
- 218. Tan, M., Le, Q.: Efficientnet: Rethinking model scaling for convolutional neural networks. In: Proceedings of International Conference on Machine Learning, pp. 6105–6114. PMLR (2019)

- 219. Tang, M., Luo, M., Zhou, J., Yang, Z., Guo, Z., Yan, F., Liu, L.: Side-channel attacks in a real scenario. Tsinghua Science and Technology 23(5), 586–598 (2018). DOI 10.26599/tst.2018.9010047
- 220. Teboh, J., Tubana, B., Udeigwe, T., Emendack, Y., Lofton, J.: Applicability of ground-based remote sensors for crop n management in sub saharan africa. Journal of Agricultural Science 4, 175– (2012). DOI 10.5539/ jas.v4n3p175
- 221. Tian, F.: An agri-food supply chain traceability system for china based on RFID & blockchain technology. In: Proceedings 13th International Conference on Service Systems and Service Management (ICSSSM). IEEE (2016). DOI 10.1109/icsssm.2016.7538424
- 222. Tilva, V., Patel, J., Bhatt, C.: Weather based plant diseases forecasting using fuzzy logic. In: Proceedings of Nirma University International Conference on Engineering (NUiCONE), pp. 1–5. IEEE (2013)
- 223. Tripathy, P.K., Tripathy, A.K., Agarwal, A., Mohanty, S.P.: Mygreen: An iot-enabled smart greenhouse for sustainable agriculture. IEEE Consumer Electronics Magazine (2021)
- 224. Uddin, M., Ayaz, M., Mansour, A., Aggoune, e.H.M., Sharif, Z., Razzak, I.: Cloud-connected flying edge computing for smart agriculture. Peer-to-Peer Networking and Applications **14**(6), 3405–3415 (2021)
- 225. Udutalapally, V., Mohanty, S.P., Pallagani, V., Khandelwal, V.: scrop: A novel device for sustainable automatic disease prediction, crop selection, and irrigation in internet-of-agro-things for smart agriculture. IEEE Sensors Journal (2020)
- 226. Uno, Y., Prasher, S., Lacroix, R., Goel, P., Karimi, Y., Viau, A., Patel, R.: Artificial neural networks to predict corn yield from compact airborne spectrographic imager data. Computers and electronics in agriculture 47(2), 149–161 (2005)
- 227. USDA, NASS: Crop Condition and Soil moisture Analytics. https://nassgeo.csiss.gmu.edu/CropCASMA/ (2021). Accessed on October 30, 2021
- 228. Van Der Werf, H.M., Zimmer, C.: An indicator of pesticide environmental impact based on a fuzzy expert system. Chemosphere **36**(10), 2225–2249 (1998)
- 229. Vangipuram, S.L.T., Mohanty, S.P., Kougianos, E.: CoviChain: A blockchain based framework for nonrepudiable contact tracing in healthcare cyber-physical systems during pandemic outbreaks. SN Computer Science 2(5) (2021). DOI 10.1007/s42979-021-00746-x
- Vangipuram, S.L.T., Mohanty, S.P., Kougianos, E.: Incentivechain: Blockchain crypto-incentive for effective usage of power and water in smart farming. In: 2022 OITS International Conference on Information Technology (OCIT), pp. 614–619 (2022). DOI 10.1109/OCIT56763.2022.00119
- 231. Vangipuram, S.L.T., Mohanty, S.P., Kougianos, E.: Cropaid: Protection of information in agriculture cyberphysical systems using distributed storage and ledger. In: D. Puthal, S. Mohanty, B.Y. Choi (eds.) Internet of Things. Advances in Information and Communication Technology, pp. 375–394. Springer Nature Switzerland, Cham (2024)
- 232. Vasisht, D., Kapetanovic, Z., Won, J., Jin, X., Chandra, R., Sinha, S., Kapoor, A.: FarmBeats: An IoT Platform for Data-Driven Agriculture. In: Networked Systems Design and Implementation (NSDI). USENIX (2017). URL https: //www.microsoft.com/en-us/research/publication/ farmbeats-iot-platform-data-driven-agriculture/

- 233. Vellidis, G., Tucker, M., Perry, C., Kvien, C., Bednarz, C.: A real-time wireless smart sensor array for scheduling irrigation. Computers and electronics in agriculture 61(1), 44–50 (2008)
- 234. Vidhya, R., Valarmathi, K.: Survey on automatic monitoring of hydroponics farms using IoT. In: Proceedings of 3rd International Conference on Communication and Electronics Systems (ICCES). IEEE (2018). DOI 10.1109/cesys.2018.8724103
- 235. Villa-Henriksen, A., Edwards, G.T., Pesonen, L.A., Green, O., Sørensen, C.A.G.: Internet of things in arable farming: Implementation, applications, challenges and potential. Biosystems Engineering **191**, 60–84 (2020)
- 236. Vincent, D.R., Deepa, N., Elavarasan, D., Srinivasan, K., Chauhdary, S.H., Iwendi, C.: Sensors driven ai-based agriculture recommendation model for assessing land suitability. Sensors 19(17), 3667 (2019)
- 237. Virk, A., Noor, M.A., Fiaz, S., Hussain, S., Hussain, H., Rehman, M., Ahsan, M., Ma, W.: Smart Farming: An Overview, pp. 191–201 (2020). DOI 10.1007/978-3-030-37794-6\_10
- 238. Vitali, G., Francia, M., Golfarelli, M., Canavari, M.: Crop Management with the IoT: An Interdisciplinary Survey. Agronomy 11(1), 181 (2021). DOI 10.3390/ agronomy11010181
- Wang, K., Zhang, S., Wang, Z., Liu, Z., Yang, F.: Mobile smart device-based vegetable disease and insect pest recognition method. Intelligent Automation & Soft Computing 19(3), 263–273 (2013)
- 240. Wang, Y., Yong, X., Chen, Z., Zheng, H., Zhuang, J., Liu, J.: The design of an intelligent livestock production monitoring and management system. In: Proceedings of IEEE 7th Data Driven Control and Learning Systems Conference (DDCLS). IEEE (2018). DOI 10.1109/ ddcls.2018.8516021
- Warden, P., Situnayake, D.: TinyML. O'Reilly Media, Inc. (2019)
- 242. Wolfert, S., Ge, L., Verdouw, C., Bogaardt, M.J.: Big data in smart farming–a review. Agricultural systems 153, 69– 80 (2017)
- 243. Wu, H., Cao, J., Yang, Y., Tung, C.L., Jiang, S., Tang, B., Liu, Y., Wang, X., Deng, Y.: Data management in supply chain using blockchain: Challenges and a case study. In: Proceedings 28th International Conference on Computer Communication and Networks (ICCCN). IEEE (2019). DOI 10.1109/icccn.2019.8846964
- 244. Wu, H.T., Tsai, C.W.: An intelligent agriculture network security system based on private blockchains. Journal of Communications and Networks 21(5), 503–508 (2019). DOI 10.1109/jcn.2019.000043
- 245. Xiao, L., Huang, W., Xie, Y., Xiao, W., Li, K.C.: A blockchain-based traceable IP copyright protection algorithm. IEEE Access 8, 49532–49542 (2020). DOI 10.1109/access.2020.2969990

- 246. Yang, F., Shu, L., Huang, K., Li, K., Han, G., Liu, Y.: A partition-based node deployment strategy in solar insecticidal lamps internet of things. IEEE Internet of Things Journal 7(11), 11223–11237 (2020). DOI 10.1109/jiot.2020.2996514
- 247. Yang, J., Shen, X., Xing, J., Tian, X., Li, H., Deng, B., Huang, J., Hua, X.s.: Quantization networks. In: Proceedings of the IEEE/CVF Conference on Computer Vision and Pattern Recognition, pp. 7308–7316 (2019)
- 248. Yang, T.J., Chen, Y.H., Sze, V.: Designing energy-efficient convolutional neural networks using energy-aware pruning. In: Proceedings of the IEEE Conference on Computer Vision and Pattern Recognition, pp. 5687–5695 (2017)
- 249. Yang, X., Shu, L., Liu, Y., Hancke, G.P., Ferrag, M.A., Huang, K.: Physical Security and Safety of IoT Equipment: A Survey of Recent Advances and Opportunities. IEEE Transactions on Industrial Informatics pp. 1–1 (2022). DOI 10.1109/TII.2022.3141408
- 250. Yao, S., Zhao, Y., Zhang, A., Su, L., Abdelzaher, T.: Deepiot: Compressing deep neural network structures for sensing systems with a compressor-critic framework. In: Proceedings of the 15th ACM Conference on Embedded Network Sensor Systems, pp. 1–14 (2017)
- 251. Yao-lin, Z., Gao-qiang, Z., Lei, Z., Jin, X.: Design of wireless multi-point temperature transmission system based on nRF24l01. In: 2011 International Conference on Business Management and Electronic Information. IEEE (2011). DOI 10.1109/icbmei.2011.5920375
- 252. Zeng, Y.: Digital music resource copyright management mechanism based on blockchain. In: Proceedings 3rd International Conference on Smart BlockChain (Smart-Block) (2020). DOI 10.1109/smartblock52591.2020.00036
- Zhang, J., Wang, Y., Li, J., Yang, P.: A neural network model for predicting cotton yields. International federation for information processing-publications-ifip 259, 1321 (2008)
- 254. Zhang, Y.: Design of the node system of wireless sensor network and its application in digital agriculture. In: Proceedings International Conference on Computer Distributed Control and Intelligent Environmental Monitoring (2011). DOI 10.1109/cdciem.2011.371
- 255. Zhao, W., Lin, S., Han, J., Xu, R., Hou, L.: Design and implementation of smart irrigation system based on lora. In: Proceedings of IEEE Globecom Workshops (GC Wkshps), pp. 1–6 (2017)
- 256. Zhou, L., Wang, L., Sun, Y., Lv, P.: BeeKeeper: A blockchain-based IoT system with secure storage and homomorphic computation. IEEE Access 6, 43472–43488 (2018). DOI 10.1109/access.2018.2847632
- 257. Zhou, Y., Moosavi-Dezfooli, S.M., Cheung, N.M., Frossard, P.: Adaptive quantization for deep neural network. In: Proceedings of Thirty-Second AAAI Conference on Artificial Intelligence (2018)